A Coarse Taxonomy of Artificial Immune Systems

Jason Brownlee jbrownlee@ict.swin.edu.au

Complex Intelligent Systems Laboratory (CIS) Centre for Information Technology Research (CITR) Faculty of Information and Communication Technologies (ICT) Swinburne University of Technology

> **Technical Report** TR2006-01 May 2006

Table of Contents

1	INTRODUCTION	.3
2	ARTIFICIAL IMMUNE SYSTEM	.3
	 2.1 Definitions 2.2 Overviews of the field	.4
3	NEGATIVE SELECTION PARADIGM	.5
4	CLONAL SELECTION PARADIGM	.7
5	IMMUNE NETWORK PARADIGM	.8
6	DANGER SIGNAL PARADIGM	10
7	OTHER WORKS	10
8	DISTRIBUTED ARTIFICIAL IMMUNE SYSTEMS	11

1 Introduction

This work provides a high-level taxonomy of the field of Artificial Immune Systems (AIS). Specifically, the taxonomy presented is not application focused as is the case in de Castro and Timmis' seminal reference on the field [212], rather the field is presented firstly by computational paradigm, and secondary by so-called thrust of research. This secondary aggregation involves the subjective organisation of work based on authors involved and the theme of the presented research. The intent of this work is to provide both a high-level overview of the principle computational paradigms of the field and provide some indication of the major works and their relationships within each paradigm. This work is not a complete bibliography, rather it contains seminal and influencial works from each selected paradigm. Section 2 considers the general field of Artificial Immune Systems, and provides references regarding definitions, overview of the field and conceptual frameworks. Section 3 considers those works inspired by the self-nonself principle refered to as the Negative Selection *Paradigm.* Section 4 considers those works inspired by Burnet's clonal selection theory refered to as the Clonal Selection Paradigm. Section 5 considers those works inspired by Jerne's immune network theory refered to as the Immune Network Paradigm. Section 6 considers those works inspired by Matzinger's danger theory refered to as the Danger Signal Paradigm. Section 7 considers those works that do not fit neatly into the selected organisation of the field. Finally, Section 8 considers works that focus on the distributed concerns of the immune system.

2 Artificial Immune System

2.1 Definitions

The field of Artificial Immune Systems (AIS) is primarily concerned with the development, demonstration, and implementation of computational tools inspired by principles and processes of the vertebrate immune system. The field of artificial immune systems is not concerned with the theoretical or computational modelling of the biological system. Standard definitions of both the field of AIS and what comprises an artificial immune system follow.

The authors de Castro and Von Zuben [210] define an AIS as

"... a computational system based upon metaphors of the biological immune system".

They go on to define Immune Engineering (IE) as

"... a meta-synthesis process that uses the information contained in the problem itself to define the solution tool to a given problem, and then apply it to obtain the problem solution".

The distinction is the IE is the engineering process, differentiated as to the conventional engineering process that uses ideas from biology, in this case an artificial immune system. Immune Engineering was later formalised by de Castro and Timmis [212]. A set of three principles was introduced that a framework for developing artificial immune system algorithms should contain, as follows:

- 1. A representation for the components of the system
- 2. A set of mechanisms to evaluate the interaction of individuals with the environment and each other
- 3. Procedures of adaptation that govern the dynamics of the system

A further, more concise definition of AIS is provided in [212] as

"... adaptive systems, inspired by theoretical immunology and observed immunological functions, principles and models, which are applied to problem solving".

This definition clearly highlights the role of theoretical immunology and immunological observations as an inspiration and not the pursuit of research in AIS. Another perspective of AIS is provided by one of the pioneers in the field Ishida [353], who defines an Immunity-Based System (IMBS) as a:

"... self-maintenance systems learned from and inspired by the immune system"

His IMBS is described to posses the following three properties:

- 1. a self-maintenance system with monitoring not only of the nonself but also of the self
- 2. a *distributed system* with autonomous components capable of mutual evaluation
- 3. an *adaptive system* with diversity and selection

Finally, Tarakanov et al. [15] introduce yet another interpretation on the field; Immunocomputing (IC) as

"... a new computing approach that replicates the principles of the information processing by proteins and immune networks".

Immunocomputing is a protein and immune network approach that attempts to address the shortcomings in AIS, specifically the lack of a proper mathematical basis and a lack of an analogous hardware platform.

2.2 Overviews of the field

This section lists works that provide an introductory reference to the field of AIS. References are listed in no particular order.

Summary	References
AIS as a soft computing paradigm	[215]
Popular and generally useful (in nature)	[83]
Usefulness of AIS as a computational intelligence paradigm	[277]
Tech reports by de castro	Introduction [209],
	Applications [210]
AIS in the last 5 years	[72]
Review paper – a survey	[44]
Overview of AIS – chapter	[168]
General overview of AIS's usefulness as a learning paradigm	[349]
Intro to the field of AIS	[71]
AIS for optimization, a survey	[344]
An overview of AIS as intrusion detection (natural application)	[328]
Application of AIS	[78]
AIS as a new learning paradigm	[349]
AIS as a pattern recognition paradigm	[213]
The usefulness of the paradigm	[277]
The importance of AIS in the next 10-15 years	[308]
Why computer scientists and immunologists should work together	[257]

2.3 Conceptual Frameworks for AIS

This section lists some references that provide conceptual frameworks for considering AIS. References are provided in no particular order.

Summary	References
Conceptual framework for AIS	Framework [310]
	Application of framework [153]
	Seeking a more complete model [241]
	More on the theme [309]

Immune Engineering	[212],
(conceptual model for engineering with AIS)	Claimed to be incomplete [310],
	More on immune engineering [202]
Related to AIS's place in optimisation	[20]
Next generation AIS	[256]
Revisiting the foundations of AIS	[14]
AIS as a soft computing paradigm	[215]
Related to problems - the need to go back and look at the roots of	[166]
the field	
Immunocomputing	[16]
Innate immunity (a framework)	[134]

3 Negative Selection Paradigm

This section summarises seminal works on the negative selection paradigm. References are listed in no particular order.

Approach	Description	References / Application
(Forrest, Smith, et al.)	Early pattern recognition work using GA's (prelude to NSA)	GA to evolve cooperative antigen-antibody model or coverage [266,267] Confirmatory work [290] Related (?) [288]
Change Detection (Negative Selection Algorithm) (NSA) (Forrest, D'haeseleer, et al.)	The original change detection algorithm, now commonly referred to as the negative selection algorithm.	Random binary strings, disassembled SPARC binaries (the instructions of which were mapped to an ASCII alphabet), and binary strings from DOS (.com) binary files [289] Formalisation, modelling and improvements [252,253]. // D'haeseleer has a few tech reports and unpublished works Overview of application to computer security [295]
Process Homeostasis (pH) (Forrest, Somayaji, et al.)	Anomaly detection in sequences of system calls known as deviation from process homeostasis (an intrusion detection system as a patch to the Linux kernel)	Sequences of system calls in privileged (root) UNIX processes [293], [301] process monitoring with intervention embedded in the Linux kernel [32] overview [33], [220] The need for diversity in computer security (related) [287] Culminating work [31] Hybridising pH and Cfengine [197]
Lightweight Intrusion Detection System (LISYS) (Forrest, Hofmeyr, et al.)	Anomaly detection in TCP/IP connections, an implementation of ARTIS for computer security	Host-based and network based intrusion detection (anomaly detection) [303], [299], [294], [292] Related (?) [304] Culminating work [302] Overview of LISYS [291] Renamed to LISYS, an implementation of ARTIS [300]
(Balthrop, Forrest, et al.)	Simplified LISYS	Confirmation and extension of a simplified LISYS [126], [196]
Artificial Immune System (ARTIS) (Hofmeyr, Forrest)	The previous work on change detection, process homeostasis, and Hofmeyr's AIS were unified (retrospectively) by Hofmeyr under the ARTIS framework.	Definition of ARTIS [300]
Computer Virus Immune System (CVIS) (Harmer)	Computational immune system for virus detection	Proposal and culminating work [255] Summary and analysis of algorithm [105]
Computer Defense Immune System (CDIS) (Williams, Gunsch, Lamont et al)	A multi-agent, hierarchical, distributed computational immune system for intrusion detection An extension of CVIS with network- based intrusion detection Uses negative selection, affinity maturation and co-stimulation	Proposal and development [254], [251], [199], [198]
Online Negative Database (Esponda, Forrest, et al.)	The concept of a negative database, which stores records that represent the complement of the data	Introduction to the concept, and demonstration of the computational complexity of the database (feasibility) [97,100] Extension of early work, again highlighting the difficulty of deriving the original data from a negative representation [98] Theoretical model (crossover closure) [99]

		Culminating work [96]
		A formal framework for positive and negative
		selection [101]
		// there are a few technical reports as well
Immunological Electronics	Fault detection system for state machine	Introduction to idea [21], [55]
(Immunotronics)	(FSM) hardware systems	Combining immunological and embryonic ideas
(Bradley, Tyrrell, et al.)		[54]
		Hardware immune system [56], [57], [58], [50]
		Related (?) [51], [262]
		Perhaps related thesis (?) [200]
		Related (?) [263]
		Perhaps an innate immunity related (?) [345]
Dynamic Clonal Selection	Dynamic clonal selection (actually a T-	Negative selection shows scaling problems, should
(DynamiCS)	cell inspired negative selection approach)	be used as a filter for invalid detectors and not
(Kim, Bentley, et al.)	algorithm for anomaly detection of real-	generation of useful detectors [185]
	time network traffic	Negative selection with niching for intrusion
		detection [184]
		An investigation of static clonal selection with a
		negative selection operator [186]
		(their IDS, but not NS (?) [183], [181])
		Proposal of the approach (DynamiCS) [189]
		Extension to the original approach [187], [182],
		[188]
		Culminating work [190]
Real-Valued Negative	Anomaly detection and classification	Experimentation and limitations of with binary
Selection	with a real-valued negative selection	matching rules [93]
(RNS, RRNS)	algorithm	Combining negative selection with a neural
(Gonzalez, Dasgupta, et al.)		network (Self-Organising Map) [92]
		Introduced RNS on classification datasets [91]
		Real-valued negative selection for anomaly
		detection, using a GA for detector generation [67],
		[90]
		Extension of work [129]
		Randomised RNS [94]
		Real-valued anomaly detection [88]
		Culminating work [89]
Variable-sized Real-valued	Extension of real-valued negative	Variable sized real-valued detectors [358], [357]
Negative Selection	selection with variable radius detectors	
(Ji, Dasgupta)		
(Dasgupta, Forrest)	Anomaly detection in time series data	Application to tool breakage detection [46]
		Application to cutting dynamics of a milling
		operation and a synthetic signal [70]
		Similar, but different authors (?) [227], [229]
(Wierzchon, et al.)	Models and improvements to the binary	Generating an optimal non-redundant pool of
	negative selection algorithm	detectors [281], [282], [280]
Information immune system	Information filtering (collaborative	Introduction to the idea of information filtering
(Chao, Forrest, et al.)	filtering) using negative selection to	with negative selection [61], [63]
	censor undesirable data	Information filtering of art (biomorphs) [62]
	(Is this an application of ARTIS?)	Use of negative preferences applied to radio for a
		group of users [60]
		// I think there is a tech report for adaptive radio
(Stibor, Timmis, et al.)	Negative selection models are too	Generation of all possible r-chunk detectors not in
	simplistic (naive)	the self set [322]
		Negative selection is too naïve (binary model), not
		appropriate for network-based IDS [320]
		Comparison of variable sized RNS with SVM [323]
		Further comparisons of real-valued negative
		selection to statistical techniques [321]
		Stibor's Phd Thesis [319]
	L	

The following lists additional uncategorised negative selection works:

- Hightower's work on gene libraries [269]
- Maths of the matching model (change detection) [265]
- Maths and introduction of r-contiguous bits [130], [131]
- Some random classification approach with positive and negative selection [102]
- On the usefulness of the NS paradigm [277]
- Extension of LISYS (synapse) [48]
- Fault detection of temperature data in a refrigeration system [53]
- Investigation into different means of generating detectors (de castro and timmis) [230]

- Image classification of pictures of wooden kitchen components with negative selection [285]
- On the use of negative selection in an AIS (unable to get...) [232]
- Application of NSA to multi-dimensional sensitive personal data (data integrity) [45]
- Application of NSA to files, system calls and time series [274]
- Image segmentation application of a negative selection approach [49]
- There may be some related work on combining pH with cfengine (Burgess)
- Negative selection application to steganography (writing a cipher) [132]
- Aircraft fault detection using negative selection (real-valued negative selection) (MILD) [43]
 - Negative selection for aircraft fault detection (ultimately real-valued negative selection...) [69]

4 Clonal Selection Paradigm

This section lists seminal works in the clonal selection paraidm. Works a listed in no particular order.

Approach	Description	References / Application
Clonal Selection Algorithm	Implementation of the clonal selection	Introduction (CSA) and application to binary character
(CLONALG)	principle and affinity maturation	recognition, TSP and function optimisation [204],
(de Castro, Von Zuben)		[209]
(Watkins, et al.)		Renamed to CLONALG [208]
		Parallel implementation of CLONALG [30]
Adaptive Clonal Selection	Extensions and simplifications to	Comparison of clonal selection and ES on dynamic
(ACS)	CLONALG	function optimisation [146]
(Walker, White, Garrett)		Improvement and application of CLONALG to binary
		character recognition [143]
		Introduction of real-valued, parameter-free adaptive
	· · · · · · · · · · · · · · · · · · ·	clonal selection (ACS) for optimisation [276]
Artificial Immune	A supervised learning clonal selection	Introduction to the technique, application to
Recognition System	algorithm for classification, derived	classification [19], [17], [29], [28]
(AIRS)	from RLAIS/AINE	Simplification of the approach (AIRS2), and applied
(Watkins, Boggess,		to classification [26]
Goodman, Timmis, et al.)		Applied to multi-class data, compared to LVQ [74]
(Greensmith)		Issues that affect the performance of AIRS [106] Investigation into the source of power of the AIRS
		algorithm (maintenance of memory cells) [75]
		Comparison between AIRS and LVQ on datasets with
		irrelevant features [228]
		Investigation of the memory cells [73]
		Parallel version of AIRS [27]
		The use of non-Euclidean distance measures [136]
		Culminating work [18]
		Application of AIRS to hierarchical multi-class
		document classification [174], [172]
B-cell Algorithm	Clonal selection approach to	Seminal (?) [157]
(BCA)	multimodal function optimisation	Related (?) [20]
(Kelsey, Timmis, et al.)	-	Application to dynamic multimodal function
-		optimisation [156]
		Culminating work (?) [155]
		Related culminating work (?) [35]
		Comparison of opt-aiNET, BCA, and a GA [160]
		Markov chain model of BCA [76]
		Theretical convergence proof – empirical [250]
Multi-objective immune	A clonal selection approach for multi-	Introduction to approach [36]
system algorithm	objective optimisation	Extension (more relaxed) to approach compared to
(MISA)		other genetic approaches [243]
(Cruz Cortes, Coello Coello,		Proof of convergence of MISA [233]
et al.)	Diagan anna ak fan agarking (11	Culminating work (?) [244]
Simple Immune Algorithm	Binary approach for combinatorial	Introduced and applied to combinatorial optimisation
(SIA) (Cutalla Nicosia)	optimisation	(?) [338] Summery [240]
(Cutello, Nicosia)	Hybridized oppress h of slope	Summary [340]
Cloning, Information Gain,	Hybridised approach of clonal	Introduction (?) [339] Application to the graph colouring problem (?) [241]
Aging (CLIGA, CLIGA+,	selection and information gain.	Application to the graph colouring problem (?) [341] Summary and extension (CLIGA+) and introduction
READI-Alg)		of a diffuse version – Reaction Diffuse Immune
KEADI-AIg)		or a unruse version – Keacuon Diffuse minimune

(Cutello, Nicosia)		Algorithm (READI-Alg) [340]
Immune Algorithm	Clonal selection approach with hyper-	Application to the HP protein folding problem [343],
(IA)	macro-mutations	Introduction (?) [335]
(Cutello, Nicosia, Pavone, et		Evaluation of different hypermutation operators (?)
al.)		[342]
		Perhaps a culminating work (?) [108]
Optimised Immune	Hard to see the difference	Introduction to the approach and comparison to
Algorithm	(extension of IA?)	CLONALG [336]
(opt-IA)		A more detailed comparison of opt-IA and
(Cutello, Nicosia, Pavone, et		CLONALG (?) [337]
al.)		Comparison to other approaches on function
		optimisation [333]
Optimised Immune	An extension of opt-IA	Introduction to approach [334]
Algorithm		
(opt-IMMALG)		
(Cutello, Nicosia, Pavone, et		
al.)		

The following lists additiona unclassified clonal selection works:

- Early work by Fukuda using GA's
 - Using a GA as the clonal selection bit (?) [326]
 - Application to resource allocation problem (?) [201]
 - Function optimization (?) [327]
- In evolving gene libraries for antibodies, Hightower drew a similarity between clonal selection and the Baldwin effect of life-time learning [270]
- Early work (?) [264]

5 Immune Network Paradigm

This section lists seminal works in the immune network paradigm. References are listed in no particular order.

Approach	Description	References / Application
(Farmer, Packard, Perelson)	Claimed to be a continuous immune network model	A dynamic model of the immune system based on the immune network theory, and similarity to classifier systems [127]
(JISYS) (Hunt, Cooke, et al.)	Learning system inspired by the immune network theory. // refers to work as content- addressable memory // similar to CBR systems	Related (?) used to classify DNA sequences [59], [149] Extension and application to the game of tic-tac-toe [148] Related (?) on the matching algorithm (?) [150] Perhaps related (?) Immunizing against fraud [147] Renamed to JISYS (?) [152] Augmented JISYS and application to learn trends in loan application data [154]
Artificial Immune Network (AIN) (Timmis, Neal, Hunt)	An extension of Hunt, Cooke's work (JISYS) with the major addition being ARB's. Immune network approach for data analysis, unsupervised clustering and visualisation	Proposal of approach for unsupervised clustering and visualisation [165] Extension and comparison with SOM [164] // related (?) [151]
Resource Limited Artificial Immune System (RLAIS) (Timmis, Neal)	An extension and rename of AIN, AIN refers to the model constructed by the approach	Application to unsupervised clustering and visualisation [163], [162] Culminating work [169]
Artificial Immune Network (AINE) (Knight, Timmis)	An extension and rename of RLAIS // proposed to be a discrete immune network model	Summary of work [167] Improvement and corroboration of results and additional testing on larger datasets [315] Investigation into parameters of AINE [158] (tech report) Assessing the performance of AINE [316] (tech report) Proposal of approach as a redesigned multilayered version of AINE (parallel to SSAIS) [317], [318] Thomas' thesis???
Self-Stabilising Artificial Immune System (SSAIS) (Neal)	An extension of RLAIS that is capable of finding stable clusters (parallel work to AINE and multi-layered redesign of AINE)	Proposed and applied to data analysis [239] Simplification of the approach [240]

Artificial Immune Network Visualisation (aiVIS) (Timmis)	A tool for visualising models prepared by AIN, RLAIS, AINE and SSAIS	Visualisation approach for unsupervised cluster models (AIN) [159]
Enhanced/Fuzzy Artificial Immune Network (Enhanced/Fuzzy AINE) (Nasraoui, Gonzalez, Dasgupta, et al.)	Extension of AINE that incorporates fuzzy sets in the ARB's	Proposal of Enhanced AINE and application to web- profile learning [247] Proposal and application of fuzzy AINE to learning user web-profiles [249] Application to synthetic data and web profiles [246]
Artificial Immune Network	Combines clonal selection and	Extension that includes a dynamically weighted B-cell (D-W-B-cell) mechanism [248] Proposal for unsupervised learning (clustering) [205]
Algorithm (aiNet) (de Castro, Von Zuben, Timmis, et al.)	immune network properties. The aiNet is proposed as an immune network extension to CLONALG. // proposed to be a discrete immune	Presentation and investigation for unsupervised data clustering, compared to SOM, parameter sensitivity [206] Applied to machine learning and data compressing
	network model	applications & comparison to neural networks [207] Summary of aiNet (and CLONALG) [213] Extracting classes and class hierarchies as well as convergence properties of the approach [214]
		Application of aiNet to bioinformatics [107]
Optimised Artificial Immune Network Algorithm (opt-aiNet) (de Castro, Von Zuben,	An extension of aiNet for function optimisation	Proposed extension (opt-aiNet) for multimodal function optimisation [211] Summary of aiNet and opt-aiNet [203] Verification of results of opt-aiNet, minor fixes [161]
Timmis, et al.)		Comparison of opt-aiNET, BCA and a GA [160] Extension of opt-aiNet (dopt-aiNet) for dynamic function optimisation [95]
Antibody Network (ABNET) (de Castro, Von Zuben, et al.)	A neural network approach inspired by the immune system (?)	Use of CLONALG to explain how the model is activated [216]
(Wierzchon, Kuzelewska)	AIN as a unification of ideas from AINE and aiNet	Proposed as a unified AIN from AINE and aiNet [283]
Simple Artificial Immune System (SAIS , YSAIS) (Gaspar, Collard, et al.)	Immune network approach for time- dependant optimisation. Makes use of clonal selection principles as well.	Proposal for time-dependant optimisation (TDO) [12] Renamed (YSAIS) more robust approach for TDO with niching-like properties [13], [11]
Recommender System (Cayzer, Aickelin) (Morrison, Aickelin) (Aickelin, Chen)	Immune network inspired (unsupervised) recommender system – collaborative filtering (given a film, make a prediction for other like films)	Proposed approach and applied to film user profiles [297] Extension of the recommender system to better understand the idiotypic effect [296] Improved version of the approach [298] Extension of (inspired by) approach user profiling of web data [325], [324] Extension of previous work, the effects of many different similarity measures, applied to movie recommendations [331], [330]
Genetic Immune Recruitment Mechanism (GIRM) (Bersini, et al.)	Recruitment strategies to improve the local search capabilities of a genetic algorithm	Related (?) [124] Related (?) [118] Immune recruitment mechanism (GIRM) (?) [122] Related to recruitment learning (?) [120] Related (?) [119] Related (?) [122] Adaptive distributed control and the recruitment mechanism applied to so control problems [123] Related book chapter (?) [121]
Distributed diagnosis model (Ishida, Mizessyn, et al.)	A distributed diagnosis model inspired by the immune network Self organisation (?) A complex systems and knowledge processing perspective using the immune network model	Related – sensor diagnosis (?) [354] Related – process diagnosis (?) [347] Distributed diagnosis application of faulty sensors in a cement plant [85] Extension including the use of an agent-based architecture based on local memory applied to an adaptive disturbance neutraliser [355] Sensor-based diagnosis using self organisation [348], [352] Related (self organisation) (?) [350]

The following lists additional uncategorised works from the immne network paradigm:

- An overview (taxonomy) of AIN approaches [171]
- Immunity-Based Genetic Algorithm: early application to VLSI clonal selection and GA [125] (IGA)
- Combining immune network with LVQ (distributed diagnosis system) [231]

- Relation to ANN
 - o Immune network as parallel-distributed processing (PDP) [86]
 - Proposal of the immune network as a PDP model (Ishida) [346]
 - Immune network as a neural network (?) [87]
 - A neural network analogy of the neural network [104]
- A confirmation and extension of Farmer's model [103]
- An extension of Farmer's model as a probablistic model [170]
- Immune network models are too simple (epitope and paratope) (Garrett) [275]

6 Danger Signal Paradigm

This section lists works from the danger signal paradigm. References are listed in no particular order.

Approach	Description	References / Application
(Aickelin, Cayzer, et al.)	Proposal of Danger theory in AIS,	Proposal of Danger Theory as a useful process in
	adoption in intrusion detection - an	AIS [332]
	alternative perspective to self/nonself	Extension of previous work, proposal as an IDS
	discrimination	[329], [176], [133]
		The use of sage and danger signals applied to
		anomaly detection in system calls [180]
(Secker, Freitas, Timmis)	A danger area is defined as an area	Introduction and proposed application as an
	(document) of interest (interesting	adaptive mailbox filter (email filter) [24]
	signal)	Proposed as an extension to AISEC [25]

7 Other Works

This section lists works that do not neatly fit into the prevous computational paradigm categories. References are listed in no particular order.

Approach	Description	References / Application
Co-evolutionary Sparse	Combining ideas from immunology and	Proposal of COSDM, evolved using a co-
Distributed Memories	Sparse Distributed Memories (SDM), an	evolutionary GA, applied to data clustering in time
(COSDM)	extension of the argument that	series data [79]
Self-Organising Sparse	immunological memory is associative	Extension called (SOSDM) applied to clustering
Distributed Memory	presented in [64]	benchmark data as well as non-stationary datasets
(SOSDM)	-	[81], [80]
(Hart, Ross)		Culminating work [77]
(Oda, White)	Spam email filter	Proposed approach [312]
	Involved creating a library of regular	Improving the accuracy [313]
	expressions (lymphocytes)	Summary and comparison [314]
		Culminating work [311]
Artificial Immune System	Spam email classification system	Proposal of AISEC as an email classification
for Email Classification		system [23]
(AISEC)		A danger theory inspired extension to AISEC [25]
(Secker, Freitas, Timmis)		Perhaps a related thesis (?) [144]
()		Secker's thesus [22]
IBM AntiVirus	Classical work by IBM on an AIS as an	Proposal of an immune system for computers [139],
(Kephart, et al.)	anti-virus system	[142]
(F)		An immune system to protect computers from the
		internet [140]
		An immune system for computers [141]
Computer Immune System	Application of both positive and negative	Proposal of the approach applied to detecting
for Fraud Detection	selection for retail fraud	financial fraud [179]
(CIFD)		
(Kim, Ong, Overill)		
Multilevel Immune Learning	Pattern recognition (anomaly detection)	Proposal of the approach [65], [47]
Algorithm	with multiple detection schema, clonal	
(MILA)	expansion, and dynamic detector	
(Dasgupta, Yu, Majumdar)	generation	
(Immunos-81)	Supervised classification system inspired	Proposal of the approach [145]
(Carter)	by populations of T and B cells (clonal	
. ,	selection)	
Site Configuration Engine	The work is on network computer	Proposal of Cfengine (not immunological) [234],
(Cfengine)	security, although there is an	[238]
(Burgess, et al.)	immunological inspiration influencing	Using ideas of immunology for reliable software

	the mode	aviatama [225]
	the work	systems [235]
		A review of the immunity model used in Cfengine
		[236]
		On hybridising pH with Cfengine [197]
		Immunological properties as they relate to lazy
		evaluation and probabilistic models [237]
		// many other unrelated references on Cfengine
(Hajela, Yoo, et al.)	Immune network to handle constraints	Related (?) [260]
	within a genetic algorithm	Application of embedded immune network in a GA
		to multi-objective design and optimisation [261],
		[178]
(Hightower, Forrest, et al.)	Evolution of gene libraries	Use of a genetic algorithm to evolve immune
		system genes for a binary immune system model
		(secondary effect is organisation within the genes)
		[269], [272]
		Extension of the previous model to include Baldwin
		learning [270]
		Culminating work [271]
Tissue / Dendric Cell	Tissue models inspired by the innate	Algorithm inspired by dendritic cells (between
Algorithm	immune system, dendric cells and toll-	innate and acquired systems) applied to
(libtissue, DCA)	like receptors	classification [177]
(Greensmith, Aickelin, et	1	Proposal for using tissue in AIS (two tissue
al.)		growing algorithms are introduced) for
		classification [258]
		Tissue models for feature extraction and clustering
		(based on SOM and toll-like receptors) [128]
		DC Algorithm (DCA) applied to classification data
		and an anomaly detection problem in real-time
		[173]
		DC Algorithm (DCA) applied to a real-time port
		scan problem [175]
		Introduction to the libtissue framework [135]

Some additional uncategorised works are listed as follows.

-

- Associative memory (CBR, instancse-based, content-addressable memory, etc)
 - Early work on associative memory in the immune system (?) [34]
 - o Early work on associative memory in an immune network [84]
 - The relation between AIS and CBR is discussed in Chao and Forrest's work [63]
 - o Also discussed in Hunt and Cooke's work

8 Distributed Artificial Immune Systems

This section considers distributed artificial immune systems, irrespective of their base computational paradigm. References are listed in no particular order.

Approach	Description	References / Application
(iNexus, iNet)	Decentralised policy negotiation -	iNet application framework/architecture [194]
(Suzuki, Yamamoto)	allowing applications to adapt their	Summary of iNexus [193], [195]
(Lee, Suzuki, et al.)	configurations to meet their needs	Related (?) [192]
	// Decentralised and autonomous policy	Culminating work [191]
	negotiation between applications within	An extension of iNet for adaptive network
	an application server?	applications (NetSphere plug-in) [39], [41], [40]
	// immune network	
	// inet can interact with inexus	
	// extension of inet as a netsphere plugin	
(Segel)	The immune system as a	Immune system as a prototype autonomous
	decentralised/distributed autonomous	decentralised system (a feedback model) [217]
	system	Related (?) (Segal's chapter in Dasgupta's book)
		[225]
		Related (?) (Segal's book) [224]
		Related (?) (Segal's chapter in his book) [220]
		Diffuse feedback system [220], [218]
		Further related diffuse feedback system - diffuse
		information network [219]
		Models of an adaptive immune system based on

(Ishida)	The immune system (immune network theory) as a model for Autonomous Decentralised Systems (ADS) // very likely to have links to his	feedback (good differential equations) [226] All about the role of feedback in the immune system [222], [221] Review of spatio-temporal models of immunology [223] Proposal of the immune network as a model for autonomous distributed systems, and the internet as an example [351]
	distributed sensor immune network research	
(Le Boudec, Sarafijanovic)	Misbehaviour detection in mobile ad hoc networks (message routing). Uses ideas of negative selection and danger signals // uses danger signals	Proposal of the approach applied to dynamic source routing (tech report) [137], [138] An extension that introduces a virtual thymus, clustering and danger signals [278] Extension with the use of a secondary response [279] // more technical reports
Immunology-derived Distributed Autonomous Robotics Architecture (IDARA, Kilorobot) (Singh, Thayer)	Coordinate and control large scale distributed robot teams	Proposal of the architecture applied to a robot mine field [305] Introduction of IDARA for the manipulation of kilorobots (large multi-robot colonies) applied to mapping scenarios [273], [307] Extension applied to large colonies of kilorobot's for a search and rescue application[306]
(Immunobot , Immunoid) (Ishiguro, Kondo, Watanabe, Uchikawa, et al.)	Autonomous robot control inspired by the immune network theory (consensus making system) // action selection mechanism // originator of the term meta-dynamics in AIS???	Autonomous robot control applied to robot navigation [8] Proposed decentralised consensus making system (Immunobot) [2] An extension applied to garbage collection with self sufficiency (Immunoid) [6], [4], [5] Related (?) (seminal Immunoid?) [3] Application to goal seeking and obstacle avoidance [7] Extension and inclusion of meta-dynamics, applied to a predator-prey simulation [1] Related (?) (book chapter) [356]
General Suppression Framework (Lau, Wong, et al.) (Ko, Lau, Lau)	Distributed control framework	Related (?) (Control system for warehouse vehicle control) [109] Related (?) [113] Related (?) [9] Automated transport system [111] Related (?) (distributed multi-agent system) [110] Distributed control framework for a modular self- reconfigurable robot arm [10] Autonomous distributed vehicle control system for an automated warehouse [114] Mathematical model of the control framework for automated warehouses, distribution centres, and automated material handling systems [112]
Distributed Autonomous Robotics System (DARS) (Meshref, VanLandingham)	Distributed autonomous system (multi- agent system)	Application of DARS to the dog and sheep problem (classic agent problem) [117] Related thesis (?) [116]
Artificial Immune System based Intelligent Multi Agent Model (AISIMAM) (Sathyanath, Shin)	Multi-agent system model	Proposal of the agent model, application to a mine detection and diffusion problem [286]
(Dasgupta)	Collaborative multi-agent decision support system // comments on the recirculation of lymphocytes	Proposal of a collaborative multi-agent decision support system as an OO platform with visualisation tools [42]

The following lists some additional uncategorised distributed AIS works.

- Use of negative selection for constraint handling in a GA island population model [38]
 - There may be more work that exists on this (defiantly on the use of NSA within a GA for constraint handling) (culminating [37])
- Immunity based learning for distributed search and constraint relaxation (?) [115]

- AIS as a context aware ubiquitous distributed system (immune network) [259]
- A distributed framework for multiple search algorithms (applied to FPGA) [153]
- Distributed algorithm for searching P2P networks (ImmuneSearch) [245]
- Multi-agent system for breaking up a problem (distributed problem solving) (immune network and MHC) applied to TSP (n-TSP) [242]
- An agent-based intrusion detection system (Dasgupta, Gonzalez) [66]
 - A very similar piece of work on agent-based intrusion detection by the same author [68]
- Robot control (autonomous, distributed, etc...)
 - Realisation of cooperative strategies in group behaviour (?) [52]
 - Immune network robot architecture for robot learning (?) [82]
 - Autonomous robot control for garbage collection (immune network) [268]
 - Immune system and fault tolerant computing (?) [284]

References

- [1] Akio Ishiguro, Toshiyuki Kondo, Yuji Watanabe, Yasuhiro Shirai, and Yoshiki Uchikawa, "Emergent construction of artificial immune networks for autonomous mobile robots," *IEEE International Conference on Systems, Man, and Cybernetics - 'Computational Cybernetics and Simulation'*, Orlando, FL, USA, pp. 1222-1228, 1997.
- [2] Akio Ishiguro, Toshiyuki Kondo, Yuji Watanabe, and Yoshiki Uchikawa, "Dynamic behavior arbitration of autonomous mobile robots using immune networks," *IEEE International Conference on Evolutionary Computation*, Perth, WA, Australia, pp. 722-727, 1995.
- [3] Akio Ishiguro, Toshiyuki Kondo, Yuji Watanabe, and Yoshiki Uchikawa, "Immunoid: An Immunological Approach to Decentralized Behavior Arbitration of Autonomous Mobile Robots," *Parallel Problem Solving from Nature IV*, pp. 666-675, 1996.
- [4] Akio Ishiguro, Yasuhiro Shirai, Toshiyuki Kondo, and Yoshiki Uchikawa,
 "Immunoid: An architecture for behavior arbitration based on theimmune networks," *Proceedings of the 1996 IEEE/RSJ International Conference on Intelligent Robots and Systems '96, (IROS 96)*, Osaka, Japan, pp. 1730-1738, 1996.
- [5] Akio Ishiguro, Yuji Watanabe, Toshiyuki Kondo, Yasuhiro Shirai, and Yoshiki Uchikawa, "A robot with a decentralized consensus-making mechanism based onthe immune system," *Proceedings. ISADS 97., Third International Symposium on Autonomous Decentralized Systems, Berlin, Germany, pp.* 231-237, 1997.
- [6] Akio Ishiguro, Yuji Watanabe, Toshiyuki Kondo, and Yasuhiro Shirai, Y. U., "Immunoid: A robot with descentralized behavior arbitration mechanisms based on the immune system," *ICMAS Workshop on Immunity-Based Systems*, 1996.
- [7] Akio Ishiguro, Yuji Watanabe, Toshiyuki Kondo, and Yoshiki Uchikawa,
 "Decentralized consensus-making mechanisms based on immune system: Application to a behavior arbitration of an autonomous mobile robot," *Proceedings of IEEE International Conference on Evolutionary Computation* (CEC'96), Nagoya, Japan, pp. 82-87, 1996.
- [8] Akio Ishiguro and Yuji Watanabe Yoshiki Uchikawa, "An immunological approach to dynamic behavior control for autonomous mobile robots," *Proceedings* 1995 IEEE/RSJ International Conference on Intelligent Robots and Systems 95 - 'Human Robot Interaction and Cooperative Robots', Pittsburgh, PA, USA, pp. 495-500, 1995.
- [9] Albert Ko, Henry Y. K. Lau, and T. L. Lau, "An immuno control framework for decentralised mechatronic control," *Proceedings of Third International Conference on Artificial Immune Systems (ICARIS 2004)*, September 13-16, 2004, pp. 91-105, 2004.
- [10] Albert Ko, Henry Y. K. Lau, and T. L. Lau, An immuno control framework for decentralised mechatronic control *International Journal of Unconventional Computing*, vol. 1, pp. 255-280, 2005.

- [11] Alessio Gaspar and Béat Hirsbrunner, "From Optimization to Learning in Changing Environments: The Pittsburgh Immune Classifier System," *Proceedings of the 1st Int. Conf. on Artificial Immune Systems (ICARIS'02)*, University of Kent at Canterbury, UK, 2002.
- [12] Alessio Gaspar and Philippe Collard, "From GAs to artificial immune systems: improving adaptation in time dependent optimization," *Proceedings of the* 1999 Congress on Evolutionary Computation (CEC 99), Washington, DC, USA, pp. 1866, 1999.
- [13] Alessio Gaspar and Philippe Collard, "Two models of immunization for time dependent optimization," *IEEE International Conference on Systems, Man, and Cybernetics,* Nashville, TN, USA, pp. 113-118, 2000.
- [14] Alex A. Freitas and Jon Timmis, "Revisiting the Foundations of Artificial Immune Systems: A Problem-Oriented Perspective," *Proceedings Artificial Immune Systems: Second International Conference, ICARIS 2003*, Edinburgh, UK, pp. 229-241, 2003.
- [15] Alexander O. Tarakanov, Victor A. Skormin, and Svetlana P. Sokolova. Immunocomputing: Principles and Applications, USA: Springer-Verlag, 2003.
- [16] Alexander Tarakanov and Dipankar Dasgupta, A formal model of an artificial immune system *Biosystems*, vol. 55, pp. 151-158, 2000.
- [17] Andrew B. Watkins, AIRS: A resource limited artificial immune classifier 2001. Mississippi State University.
- [18] Andrew B. Watkins, Exploiting Immunological Metaphors in the Development of Serial, Parallel, and Distributed Learning Algorithms 2005. University of Kent.
- [19] Andrew B. Watkins and Lois C. Boggess, "A Resource Limited Artificial Immune Classifier," *Proceedings of Congress on Evolutionary Computation*, HI, USA, pp. 926-931, May 2002.
- [20] Andrew Hone and Johnny Kelsey, "Optima, Extrema, and Artificial Immune Systems," *Proceedings Artificial Immune Systems: Third International Conference, ICARIS 2004*, Catania, Sicily, Italy, pp. 80-90, 2004.
- [21] Andrew M. Tyrrell, "Computer know thy self!: a biological way to look at faulttolerance," *Proceedings 25th Euromicro Conference (EUROMICRO '99)*, pp. 129-135, 1999.
- [22] Andrew Secker, Artificial Immune Systems for Web Content Mining: Focusing on the Discovery of Interesting Information 2006. University of Kent.
- [23] Andrew Secker, Alex A. Freitas, and Jon Timmis, "AISEC: An Artificial Immune System for E-mail Classification," *Proceedings of the Congress on Evolutionary Computation (CEC2003)*, Canberra. Australia, pp. 131-139, 2003.
- [24] Andrew Secker, Alex A. Freitas, and Jon Timmis, "A Danger Theory Inspired Approach to Web Mining," *Proceedings of the 2nd International Conference*

on Artificial Immune Systems, pp. 156-167, 2003.

- [25] Andrew Secker, Alex A. Freitas, and Jon Timmis. Towards A Danger Theory Inspired Artificial Immune System for Web Mining. In: Web mining: Applications and Techniques, ed. A. Scime. Idea Group Publishing, 2005.
- [26] Andrew Watkins and Jon Timmis, "Artificial Immune Recognition System (AIRS): Revisions and Refinements," *1st International Conference on Artificial Immune Systems (ICARIS2002)*, University of Kent at Canterbury, pp. 173-181, 2002.
- [27] Andrew Watkins and Jon Timmis, "Exploiting Parallelism Inherent in AIRS, an Artificial Immune Classifier," *Proceedings of the 3rd International Conference on Artificial Immune Systems (ICARIS2004)*, Catania, Sicily, Italy, pp. 427-438, 2004.
- [28] Andrew Watkins, Jon Timmis, and Lois Boggess, Artificial Immune Recognition System (AIRS): An Immune-Inspired Supervised Learning Algorithm *Genetic Programming and Evolvable Machines*, vol. 5, pp. 291-317, Sep, 2004.
- [29] Andrew Watkins and Lois Boggess, "A New Classifier Based on Resource Limited Artificial Immune Systems," *Proceedings of Congress on Evolutionary Computation*, Honolulu, USA, pp. 1546-1551, May 2002.
- [30] Andrew Watkins, Xintong Bi, and Amit Phadke, "Parallelizing an Immune-Inspired Algorithm for Efficient Pattern Recognition," *Intelligent Engineering Systems through Artificial Neural Networks: Smart Engineering System Design: Neural Networks*, pp. 225-230, 2003.
- [31] Anil Buntwal Somayaji, Operating System Stability and Security through Process Homeostasis 2002. The University of New Mexico.
- [32] Anil Somayaji and Stephanie Forrest, "Automated Response Using System-Call Delays," *Proceedings of the 9th USENIX Security Symposium*, pp. 185–198, 2000.
- [33] Anil Somayaji, Steven Hofmeyr, and Stephanie Forrest, "Principles of a Computer Immune System," New Security Paradigms Workshop: Proceedings of the 1997 workshop on New security paradigms, Langdale, Cumbria, United Kingdom, pp. 75-82, 1998.
- [34] C. J. Gibert and T. W. Routen, "Associative memory in an immune-based system," *Proceedings of the twelfth national conference on Artificial intelligence* (AAAI'94), Seattle, Washington, United States, pp. 852-857, 1994.
- [35] Camilla Edmonds, Artificial Immune Networks for Function Optimisation 2003. Computing Laboratory, University of Kent.
- [36] Carlos A. Coello Coello and Nareli Cruz Cortes, "An Approach to Solve Multiobjective Optimization Problems Based on an Artificial Immune System," *First International Conference on Artificial Immune Systems* (*ICARIS*), University of Kent at Catenbury, UK, pp. 212-221, 2002.
- [37] Carlos A. Coello Coello and Nareli Cruz Cortes, Hybridizing a genetic algorithm

with an artificial immune system for global optimization *Engineering Optimization*, vol. 36, pp. 607-634, Oct, 2004.

- [38] Carlos A. Coello Coello and Nareli Cruz Cortés, "A Parallel Implementation of an Artificial Immune System to Handle Constraints in Genetic Algorithms: Preliminary Results," *Congress on Evolutionary Computation (CEC'2002)*, IEEE Service Center, Piscataway, New Jersey, USA, pp. 819-824, 2002.
- [39] Chonho Lee and Junichi Suzuki, "Autonomic Adaptation of Network Applications with the iNet Artificial Immune System," *Communications, Internet, and Information Technology (CIIT),* Cambridge, USA, pp. 496, 2005.
- [40] Chonho Lee and Junichi Suzuki, "Efficient Behavior Selection for Immunologically-Inspired Agents in the NetSphere Architecture," *Proceedings of the 9th World Multiconference on Systemics, Cybernetics and Informatics (WMSCI-2005)*, Orlando, FL, USA, 2005.
- [41] Chonho Lee, P. Champrasert, and Junichi Suzuki, "iNet: A Biologically-inspired Adaptation Mechanism for Autonomic Grids," *Proceedings of the 2nd IEEE Upstate New York Workshop on Communications and Networking*, Rochester, NY, 2005.
- [42] D. Dasgupta, "An artificial immune system as a multi-agent decision support system," *IEEE International Conference on Systems, Man, and Cybernetics, San Diego, CA, USA, pp. 3816-3820, 1998.*
- [43] D. Dasgupta, K. KrishnaKumar, D. Wong, and M. Berry, "Immunity-Based Aircraft Fault Detection System," *Proceedings of American Institute of Aeronautics* and Astronautics, Chicago, USA, 2004.
- [44] D. Dasgupta and N. Attoh-Okine, "Immunity-based systems: a survey," IEEE International Conference on Systems, Man, and Cybernetics, 1997, Orlando, FL, USA, pp. 369-374, 1997.
- [45] D. Dasgupta and N. S. Majumdar, "Anomaly detection in multidimensional data using negative selection algorithm," *Proceedings of the 2002 Congress on Evolutionary Computation, (CEC '02)*, Honolulu, HI, USA, pp. 1039-1044, 2002.
- [46] D. Dasgupta and S. Forrest, "Tool breakage detection in milling operations using a negative-selection algorithm," Department of Computer Science, University of New Mexico, NM, USA, CS95-5, 1995.
- [47] D. Dasgupta, S. Yu, and N. S. Majumdar, MILA multilevel immune learning algorithm and its application to anomaly detection *Soft Computing - A Fusion of Foundations, Methodologies and Applications*, vol. 9, pp. 172-184, Mar, 2005.
- [48] D. Duncombe, G. Mohay, and A. Clark, "Synapse: Auto-correlation and Dynamic Attack Redirection in an Immunologically-inspired IDS," *Fourth Australasian Information Security Workshop (Network Security) (AISW* 2006), Hobart, Australia, pp. 135-144, 2006.
- [49] D. F. McCoy and V. Devarajan, "Artificial immune systems and aerial image segmentation," *IEEE International Conference on Systems, Man, and*

Cybernetics, 'Computational Cybernetics and Simulation', Orlando, FL, USA, pp. 867-872, 1997.

- [50] D. W. Bradley and A. M. Tyrrell, "Hardware fault tolerance: an immunological solution," *IEEE International Conference on Systems, Man, and Cybernetics*, Nashville, TN, USA, pp. 107-112, 2000.
- [51] D. W. Bradley and A. M. Tyrrell, "Multi-layered Defence Mechanisms: Architecture, Implementation and Demonstration of a Hardware Immune System," *Proceedings Evolvable Systems: From Biology to Hardware: 4th International Conference, ICES 2001*, Tokyo, Japan, pp. 140, 2001.
- [52] D. W. Lee, H. B. Jun, and K. B. Sim, "Artificial immune system for realization of cooperative strategies and group behavior in collective autonomous mobile robots," *Proceedings of the Forth International Symposium on Artificial Life* and Robotics, AAAI, pp. 232-235, 1999.
- [53] Dan W. Taylor and David W. Corne, "An Investigation of the Negative Selection Algorithm for Fault Detection in Refrigeration Systems," *Proceedings Artificial Immune Systems: Second International Conference, ICARIS 2003*, Edinburgh, UK, pp. 34-45, 2003.
- [54] Daryl Bradley, Cesar Ortega, and Andy Tyrrell, "Embryonics + Immunotronics: A Bio-Inspired Approach to Fault Tolerance," *The Second NASA/DoD Workshop on Evolvable Hardware (EH'00)*, Palo Alto, California, pp. 205-224, 2000.
- [55] Daryl W. Bradley and Andy M. Tyrrell, "Immunotronics: Hardware Fault Tolerance Inspired by the Immune System," *Evolvable Systems: From Biology to Hardware, Third International Conference, ICES*, Edinburgh, Scotland, UK, pp. 11-20, 2000.
- [56] Daryl W. Bradley and Andy M. Tyrrell, "The architecture for a hardware immune system," *Proceedings of The Third NASA/DoD Workshop on Evolvable Hardware*, Long Beach, CA, USA, pp. 193-200, 2001.
- [57] Daryl W. Bradley and Andy M. Tyrrell, Immunotronics: Novel finite-state-machine architectures with built-in self-test using self-nonself differentiation *IEEE Transactions on Evolutionary Computation*, vol. 6, pp. 227-238, Jun, 2002.
- [58] Daryl W. Bradley and Andy Tyrrell, "A hardware immune system for benchmark state machine error detection," *Proceedings of the 2002 Congress on Evolutionary Computation (CEC '02)*, Honolulu, HI, USA, pp. 813-818, 2002.
- [59] Denise E. Cooke and John E. Hunt, "Recognising promoter sequences using an artificial immune system," *Proceedings of the International Conference on Intelligent Systems for Molecular Biology, (ISMB), International Conference on Intelligent Systems for Molecular Biology*, pp. 89-97, 1995.
- [60] Dennis L. Chao, Justin Balthrop, and Stephanie Forrest, "Adaptive Radio: Achieving consensus using negative preferences," Group '05: Proceedings of the 2005 International ACM SIGGROUP Conference on Supporting Group Work, Sanibel Island, Florida, United States, pp. 120-123, 2005.

- [61] Dennis L. Chao and Stephanie Forrest, "Information Immune Systems," International Conference on Artificial Immune Systems (ICARIS), University of Kent at Canterbury, Canterbury, England, pp. 132-140, 2002.
- [62] Dennis L. Chao and Stephanie Forrest, "Generating Biomorphs with an Aesthetic Immune System," Proceedings of the Eighth International Conference on Artificial Life (Artificial Life VIII), Sydney, Australia, pp. 89-92, 2003.
- [63] Dennis L. Chao and Stephanie Forrest, Information Immune Systems *Genetic Programming and Evolvable Machines*, vol. 4, pp. 311-331, Dec, 2003.
- [64] Derek J Smith, Stephanie Forrest, and Alan S. Perelson, "Immunological memory is associative," *ICMAS '96: Multiagent systems workshop on immunity-based* systems, Kyoto, Japan, pp. 62-70, 1996.
- [65] Dipankar Dasgupt, Senhua Yu, and Nivedita Sumi Majumdar, "MILA Multilevel Immune Learning Algorithm," *Proceedings, Part I Genetic and Evolutionary Computation Conference (GECCO 2003)*, Chicago, IL, USA, pp. 183-194, 2003.
- [66] Dipankar Dasgupta and Fabio A. Gonzalez, "An Intelligent Decision Support System for Intrusion Detection and Response," *Proceedings of the International Workshop on Information Assurance in Computer Networks: Methods, Models, and Architectures for Network Security (MMM-ACNS),* St. Petersburg, Russia, pp. 1-14, 2001.
- [67] Dipankar Dasgupta and Fabio Gonzalez, An immunity-based technique to characterize intrusions in computer networks *IEEE Transactions on Evolutionary Computation*, vol. 6, pp. 281-291, Jun, 2002.
- [68] Dipankar Dasgupta and Hal Brian, Mobile Security Agents for Network Traffic Analysis DARPA Information Survivability Conference and Exposition (DISCEX II'01), vol. 2, pp. 13322001.
- [69] Dipankar Dasgupta, Kalmanjie Krishna Kumar, D. Wong, and M. Berry, "Negative Selection Algorithm for Aircraft Fault Detection," *ICARIS 2004: 3rd International Conference on Artificial Immune Systems*, Catania, Sicily, Italy, 2004.
- [70] Dipankar Dasgupta and Stephanie Forrest, "Novelty Detection in Time Series Data using Ideas from Immunology," *Proceedings of The International Conference* on Intelligent Systems, pp. 82-87, 1999.
- [71] Dipankar Dasgupta and Stephanie Forrest, "Artificial immune systems: a new frontier in artificial intelligence," *Proceedings of the Second International Conference on Intelligent Processing and Manufacturing of Materials* (*Ipmm'99*), pp. 257-268, Aug. 1999.
- [72] Dipankar Dasgupta, Zhou Ji, and Fabio Gonzalez, "Artificial immune system (AIS) research in the last five years," *The 2003 Congress on Evolutionary Computation (CEC '03)*, pp. 123-130, 2003.
- [73] Donald E. Goodman and Lois C. Boggess, "The role of hypothesis filtering in (AIRS), an artificial immune classifier," *Intelligent Engineering Systems Through Artificial Neural Networks*, pp. 243-248, 2004.

- [74] Donald Goodman, Lois Boggess, and Andrew Watkins, "Artificial Immune System Classification of Multiple-Class Problems," *Intelligent Engineering Systems through Artificial Neural Networks: Smart Engineering System Design: Neural Networks*, pp. 179-184, 2002.
- [75] Donald Goodman, Lois Boggess, and Andrew Watkins, "An Investigation into the Source of Power for AIRS, an Artificial Immune Classification System," *Proceedings of the International Joint Conference on Neural Networks* (IJCNN'03), pp. 1678-1683, 2003.
- [76] Edward Clark, Andrew Hone, and Jon Timmis, "A markov chain model of the B-cell algorithm," *Artificial Immune Systems*, pp. 318-330, 2005.
- [77] Emma Hart, Immunology as a Metaphor for Computational Information Processing: Fact of Fiction? 2002. Artificial Intelligence Applications Institute Division of Informatics, University of Edinburgh.
- [78] Emma Hart and Jonathan Timmis, "Application Areas of AIS: The Past, The Present and The Future," *Proceedings Artificial Immune Systems: 4th International Conference, ICARIS 2005*, Banff, Alberta, Canada, pp. 483, 2005.
- [79] Emma Hart and Peter Ross, "Clustering Moving Data With a Modified Immune Algorithm," *Proceedings Applications of Evolutionary Computing : EvoWorkshops, EvoCOP, EvoFlight, EvoIASP, EvoLearn, and EvoSTIM,* Como, Italy, pp. 394, 2001.
- [80] Emma Hart and Peter Ross, "Exploiting the Analogy Between Immunology and Sparse Distributed Memories: A System for Clustering Non-Stationary Data," *1st International Conference on Artificial Immune Systems*, Canterbury, UK, pp. 49-58, 2002.
- [81] Emma Hart and Peter Ross, Exploiting the Analogy between the Immune System and Sparse Distributed Memories *Genetic Programming and Evolvable Machines*, vol. 4, pp. 333-358, Dec, 2003.
- [82] Emma Hart, Peter Ross, Andrew Webb, and Alistair Lawson, "A Role for Immunology in "Next Generation" Robot Controllers," *Proceedings Artificial Immune Systems: Second International Conference, ICARIS 2003*, Edinburgh, UK, pp. 46-56, 2003.
- [83] Erica Klarreich, Inspired by immunity *Nature*, vol. 415, pp. 468-470, Jan 31, 2002.
- [84] F. Abbattista, G. Di Gioia, G. Di Santo, and A. M. Fanelli, "An associative memory based on the immune networks," *IEEE International Conference on Neural Networks*, Washington, DC, USA, pp. 519-523, 1996.
- [85] F. Mizessyn and Yoshiteru Ishida, "Immune networks for cement plants," Proceedings International Symposium on Autonomous Decentralized Systems, ISADS 93, Kawasaki, Japan, pp. 282-288, 1993.
- [86] F. T. Vertosick and R. H. Kelly, Immune network theory: a role for parallel distributed processing? *Immunology*, vol. 66, pp. 1-7, Jan, 1989.
- [87] F. T. Vertosick and R. H. Kelly, The immune system as a neural network: a multiepitope approach *Journal of theoretical biology*, vol. 150, pp. 225-237, May

21, 1991.

- [88] Fabio A. González and Dipankar Dasgupta, Anomaly Detection Using Real-Valued Negative Selection *Genetic Programming and Evolvable Machines*, vol. 4, pp. 383-403, Dec, 2003.
- [89] Fabio Gonzalez, A Study of Artificial Immune Systems Applied to Anomaly Detection 2003. The University of Memphis.
- [90] Fabio Gonzalez and Dipanakar Dasgupta, "An Immunogenetic Technique to Detect Anomalies in Network Traffic," *Proceedings of the genetic and evolutionary compution conference, GECCO 2002*, New York, pp. 1081-1088, 2002.
- [91] Fabio Gonzalez and Dipankar Dasgupta, "Combining negative selection and classification techniques for anomaly detection," *Combining negative selection and classification techniques for anomaly detection*, USA, pp. 705-710, 2002.
- [92] Fabio Gonzalez and Dipankar Dasgupta, "Neuro-immune and self-organizing map approaches to anomaly detection: A comparison," *Proceedings of the First International Conference on Artificial Immune Systems*, Canterbury, UK, pp. 203-211, 2002.
- [93] Fabio González, Dipankar Dagupta, and J. Gómez, "The effect of binary matching rules in negative selection," *Proceedings of the Genetic and Evolutionary Computation Conference (GECCO)*, pp. 198-209, 2003.
- [94] Fabio González, Dipankar Dasgupta, and Luis Fernando Niño, "A Randomized Real-Valued Negative Selection Algorithm," *Proceedings Artificial Immune Systems: Second International Conference, ICARIS 2003*, Edinburgh, UK, pp. 261-272, 2003.
- [95] Fabricio Olivetti de França, Fernando J. Von Zuben, and Leandro Nunes de Castro, "An artificial immune network for multimodal function optimization on dynamic environments," *Proceedings of the 2005 conference on Genetic and evolutionary computation (GECCO 2005)*, Washington DC, USA, pp. 289-296, 2005.
- [96] Fernando Esponda, Negative Representations of Information 2005. University of New Mexico.
- [97] Fernando Esponda, Elena S. Ackley, Stephanie Forrest, and Paul Helman, "Online Negative Databases," *Proceedings: Third International Conference Artificial Immune Systems, ICARIS 2004*, Catania, Sicily, Italy, pp. 175-188, 2004.
- [98] Fernando Esponda, Elena S. Ackley, Stephanie Forrest, and Paul Helman, On-line Negative Databases International Journal of Unconventional Computing, vol. 1, pp. 201-220, 2005.
- [99] Fernando Esponda, Stephanie Forrest, and Paul Helman, "The Crossover Closure and Partial Match Detection," *Proceedings Artificial Immune Systems: Second International Conference, ICARIS 2003,* Edinburgh, UK, pp. 249-260, 2003.
- [100] Fernando Esponda, Stephanie Forrest, and Paul Helman, "Enhancing Privacy through Negative Representations of Data," UNM Computer Science, University of

New Mexico, NM, USA, TR-CS-2004-18, 2004.

- [101] Fernando Esponda, Stephanie Forrest, and Paul Helman, A formal framework for positive and negative detection schemes *IEEE Transactions on Systems, Man* and Cybernetics, Part B: Cybernetics, vol. 34, pp. 357-373, Feb, 2004.
- [102] Francisco Azuaje, "An immune-inspired approach to learning and classification," *The* 21st IASTED International Multi-Conference on Applied Informatics (AI 2003), Innsbruck; Austria, pp. 1-6, 2003.
- [103] Francisco Martins and Neva Slani, "Computing with Idiotypic Networks," Intelligent Information Processing and Web Mining, Proceedings of the International IIS: IIPWM'05 Conference, Gdansk, Poland, pp. 673-684, 2005.
- [104] G. W. Hoffmann, A Neural Network Model Based on the Analogy with the Immune System *Journal of Theorectical Immunology*, vol. 122, pp. 33-67, Sep 7, 1986.
- [105] Gary B. Lamont, Robert E. Marmelstein, and David A. Van Veldhuizen. Chapter Eleven - A Distributed Architecture for a Self-Adaptive Computer Virus Immune System. In: *New Ideas in Optimization*, eds. David Corne, Marco Dorigo, and Fred Glover. UK: McGraw-Hill Publishing Company, 1999.pp. 167-184.
- [106] Gaurav Marwah and Lois Boggess, "Artificial Immune Systems for classification: Some issues," *First International Conference on Artificial Immune Systems*, September 2002, pp. 149-153, 2002.
- [107] George Barreto Bezerra and Leandro Nunes de Castro, "Bioinformatics Data Analysis Using an Artificial Immune Network," *Proceedings Artificial Immune Systems: Second International Conference, ICARIS 2003*, Edinburgh, UK, pp. 22-33, 2003.
- [108] Giuseppe Nicosia, Immune Algorithms for Optimization and Protein Structure Prediction 2004. Department of Mathematics and Computer Science, University of Catania.
- [109] Henry Y.K. Lau and Vicky W.K. Wong, "Immunologic Control Framework for Automated Material Handling," *Proceedings Artificial Immune Systems: Second International Conference, ICARIS 2003*, Edinburgh, UK, pp. 57-68, 2003.
- [110] Henry Y.K. Lau and Vicky W.K. Wong, "Immunologic Responses Manipulation of AIS Agents," *Proceedings Artificial Immune Systems: Third International Conference, ICARIS 2004,* Catania, Sicily, Italy, pp. 65-79, 2004.
- [111] Henry Y.K. Lau and Vicky W.K. Wong, A strategic behavioral-based intelligent transport system with artificial immune system *IEEE International Conference on Systems, Man and Cybernetics*, vol. 4, pp. 3909-3914, 2004.
- [112] Henry Y.K. Lau and Vicky W.K. Wong, An Immunity-Based Distributed Multiagent-Control Framework *IEEE Transactions on Systems, Man and Cybernetics, Part A*, vol. 36, pp. 91-108, Jan, 2006.
- [113] Henry Y. K. Lau, Vicky W. K. Wong, and Ivan S. K. Lee, Immunity-based

autonomous guided vehicles control *Design and application of hybrid intelligent systems*, vol. pp. 233-243, 2003.

- [114] Henry Y. K. Lau, Vicky W. K. Wong, and Ivan S. K. Lee, Immunity-based autonomous guided vehicles control *Applied Soft Computing*, vol. 2005.
- [115] Hideyuki Aisu and Hiroyuki Mizutani, "Immunity-Based Learning Integration of Distributed Search and Constraint Relaxation," Workshop on the Immunity-Based Systems, Second International Conference on Multiagent Systems ICMAS '96, Keihanna Plaza, Kyoto, Japan, 1996.
- [116] Hossam Meshref, Modelling Autonomous Agents' Behavior Using Neuro-Immune Networks 2002. Department of Electrical and Computer Engineering, Virginia Tech.
- [117] Hossam Meshref and H. VanLandingham, "Artificial immune systems: application to autonomous agents," *IEEE International Conference on Systems, Man, and Cybernetics*, Nashville, TN, USA, pp. 61-66, 2000.
- [118] Hugues Bersini, "Immune network and adaptive control," *Proceedings of the First European Conference on Artificial Life*, pp. 217-226, 1991.
- [119] Hugues Bersini, "The Interplay between the dynamics and the meta-dynamics of the immune network," *Proceedings of th third conference on Artificial Life*, Santa Fe, USA, 1992.
- [120] Hugues Bersini, "Reinforcement and recruitment learning for adaptive process control," *Proceedings of the international Fuzzy Association Conference* (*IFAC/IFIP/IMACS*) on Artificial Intelligence in Real Time Control, pp. 331-337, 1992.
- [121] Hugues Bersini. The Endogenous Double Plasticity of the Immune Network and the Inspiration to be drawn for Engineering Artifacts. In: Artificial Immune Systems and their Applications, ed. D. Dasgupta. Germany: Springer-Verlag, 1999.pp. 22-44.
- [122] Hugues Bersini and Francisco J. Varela, "The immune recruitment mechanism: A selective evolutionary strategy," *Proceedings of the 4th International Conference on Genetic Algorithms*, San Diego, CA, USA, pp. 520-526, 1991.
- [123] Hugues Bersini and Francisco J. Varela. Chapter 11: The immune learning mechanisms: reinforcement, recruitment and their applications. In: *Computing with Biological Metaphors*, ed. Ray Paton. London, UK: Chapman & Hall, 1994.pp. 166-192.
- [124] Hugues Bersini and Francisco Varela, "Hints for Adaptive Problem Solving Gleaned from Immune Networks," *Proceedings of the 1st Workshop on Parallel Problem Solving from Nature*, pp. 343-354, 1990.
- [125] I. Tazawa, S. Koakutsu, and H. Hirata, "An Immunity based Genetic Algorithm and its Application to the VLSI Floorplan Design Problem," *Proceedings of IEEE International Conference on Evolutionary Computation*, Nagoya, Japan, pp. 417-421, 1996.

- [126] J. Balthrop, S. Forrest, and M. R. Glickman, "Revisiting LISYS: parameters and normal behavior," *Proceedings of the 2002 Congress on Evolutionary Computation (CEC'02)*, Honolulu, HI, USA, pp. 1045-1050, 2002.
- [127] J. D. Farmer, N. H. Packard, and Alan S. Perelson, The immune system, adaptation, and machine learning *Physica D*, vol. 22, pp. 187-204, 1986.
- [128] J. Feyereisl and Uwe Aickelin, "Artificial Immune Tissue using Self-Organizing Networks," Workshop on Artificial Immune Systems and Immune System Modelling, Bristol, UK, pp. 5-6, 2006.
- [129] J. Gomez, F. Gonzalez, and D. Dasgupta, "An immuno-fuzzy approach to anomaly detection," *The 12th IEEE International Conference on Fuzzy Systems, FUZZ* '03. pp. 1219-1224, 2003.
- [130] J. K. Percus, O. E. Percus, and Alan S. Perelson, Predicting the Size of the T-Cell Receptor and Antibody Combining Region from Consideration of Efficient Self-Nonself Discrimination *Proceedings of the National Academy of Sciences*, vol. 90, pp. 1691-1695, 1993.
- [131] J. K. Percus, O. E. Percus, and Alan S. Perelson, Probability of self-nonself discrimination NATO ASI SERIES H CELL BIOLOGY, vol. pp. 631993.
- [132] Jacob T. Jackson, Gregg H. Gunsch, Roger L. Claypoole Jr., and Gary B. Lamont, Blind Steganography Detection Using a Computational Immune System Approach: A Proposal *International Journal of Digital Evidence*, vol. 4, pp. 1-19, 2002.
- [133] Jamie Twycross, "Immune Systems, Danger Theory and Intrusion Detection," *AISB* 2004 Symposium on Immune System and Cognition, Leeds, U.K, 2004.
- [134] Jamie Twycross and Uwe Aickelin, "Towards a Conceptual Framework for Innate Immunity," Proceedings ICARIS-2005, 4th International Conference on Artificial Immune Systems, Banff, Canada, pp. 112-125, 2005.
- [135] Jamie Twycross and Uwe Aickelin, "libtissue implementing innate immunity," *IEEE Congress on Evolutionary Computation (CEC06)*, Vancouver, Canada, Unpublished (Accepted, 2006).
- [136] Janna S. Hamaker and Lois Boggess, "Non-Euclidean distance measures in AIRS, an artificial immune classification system," *Proceedings of the 2004 Congress of Evolutionary Computation (CEC2004)*, pp. 1067-1073, 2004.
- [137] Jean-Yves Le Boudec and Slavisa Sarafijanovic, "An Artificial Immune System Approach to Misbehavior Detection in Mobile Ad-Hoc Networks," School of Computer and Communication Sciences, EPFL, Lausanne, Switzerland, IC/2003/59, 2003.
- [138] Jean-Yves Le Boudec and Slavisa Sarafijanovic, "An Artificial Immune System Approach to Misbehavior Detection in Mobile Ad Hoc Networks," *Revised Selected Papers: Biologically Inspired Approaches to Advanced Information Technology: First International Workshop, BioADIT 2004*, Lausanne, Switzerland, pp. 396, 2004.
- [139] Jeffrey O. Kephart, "A Biologically Inspired Immune System for Computers,"

Proceedings of the Fourth International Workshop on Synthesis and Simulation of Living Systems, pp. 130-139, 1994.

- [140] Jeffrey O. Kephart, G. B. Sorkin, and M. Swimmer, "An immune system for cyberspace," *IEEE International Conference on Systems, Man, and Cybernetics, 'Computational Cybernetics and Simulation'*, Orlando, FL, USA, pp. 879-884, 1997.
- [141] Jeffrey O. Kephart, Gregory B. Sorkin, Morton Swimmer, and Steve R. White, "Blueprint for a Computer Immune System," *Virus Bulletin International Conference*, San Francisco, California, USA, 1997.
- [142] Jeffrey O. Kephart, Gregory B. Sorkin, William C. Arnold, David M. Chess, Gerald J. Tesauro, and Steve R. White, "Biologically inspired defences against computer viruses," *Proceedings of the 14th International Joint Conference* on Artificial Intelligence (IJCA '95), Montreal, Canada, pp. 985-996, 1995.
- [143] Jennifer White and Simon M. Garrett, "Improved Pattern Recognition with Artificial Clonal Selection?," *Proceedings Artificial Immune Systems: Second International Conference, ICARIS 2003, Edinburgh, UK, pp. 181-193, 2003.*
- [144] Jeong Sik Jang, AISEC: An Empirical Investigation into an Artificial Immune System for Email Classification 2004.
- [145] Jerome H. Carter, The immune system as a model for classification and pattern recognition *Journal of the American Informatics Association*, vol. 7, pp. 28-41, 2000.
- [146] Joanne H. Walker and Simon M. Garrett, "Dynamic Function Optimisation: Comparing the Performance of Clonal Selection and Evolutionary Strategies," *ICARIS-2003*, Edinburgh, pp. 273-284, 2003.
- [147] John E. Hunt, C. King, and Denise Cooke, Immunizing against fraud IEEE Colloquium on Knowledge Discovery and Data Mining, vol. 4, pp. 1-4, 1996.
- [148] John E. Hunt and Denise E. Cooke, "An adaptive, distributed learning system based on the immune system," 'Intelligent Systems for the 21st Century'., IEEE International Conference on Systems, Man and Cybernetics, Vancouver, BC, Canada, pp. 2494-2499, 1995.
- [149] John E. Hunt and Denise E. Cooke, Learning using an artificial immune system *Journal of Network and Computer Applications*, vol. 19, pp. 189-212, 1996.
- [150] John E. Hunt, Denise E. Cooke, and Horst Holstein, "Case Memory and Retrieval Based on the Immune System," *Proceedings of the First International Conference on Case-Based Reasoning Research and Development*, pp. 205-216, 1995.
- [151] John E. Hunt and Jon Timmis, "Evolving and Visualizing a Case Database using an Immune Network," *European Conference on Artificial Intelligence (ECAI* 98), 1998.
- [152] John E. Hunt, Jon Timmis, Denise Cooke, Mark Neal, and C. King. Jisys: The Development of an Artificial Immune System for Real World Applications.

In: Artificial Immune Systems and their Applications, ed. D. Dasgupta. Germany: Springer-Verlag, 1999.pp. 157-186.

- [153] John Newborough and Susan Stepney, "A Generic Framework for Population-Based Algorithms, Implemented on Multiple FPGAs," *Artificial Immune Systems:* 4th International Conference, ICARIS 2005, Banff, Canada, pp. 43-55, 2005.
- [154] John Timmis, Mark Neal, and John Hunt, "Augmenting an artificial immune network," *IEEE International Conference on Systems, Man, and Cybernetics,* San Diego, CA, USA, pp. 3821-3826, 1998.
- [155] Johnny Kelsey, An Immune Inspired Algorithm for Function Optimisation 2004. Computing Laboratory, University of Kent Canterbury.
- [156] Johnny Kelsey, J. Timmis, and A. Hone, "Chasing chaos," *The 2003 Congress on Evolutionary Computation*, (CEC '03), pp. 413-419, 2003.
- [157] Johnny Kelsey and Jon Timmis, "Immune Inspired Somatic Contiguous Hypermutation for Function Optimisation," *Proceedings, Part I Genetic and Evolutionary Computation Conference (GECCO 2003)*, Chicago, IL, USA, pp. 207-218, 2003.
- [158] Jon Timmis, "On parameter adjustment of the immune inspired machine learning algorithm AINE," Computing Laboratory, University of Kent at Canterbury, Canterbury, Kent, UK, 12-00, Nov 2000.
- [159] Jon Timmis, "aivis Artificial Immune Network Visualisation," Conference Proceedings of EuroGraphics, University College London, UK, pp. 61-69, 2001.
- [160] Jon Timmis, C. Edmonds, and Johnny Kelsey, "Assessing the Performance of Two Immune Inspired Algorithms and a Hybrid Genetic Algorithm for Function Optimisation," *Proceedings of the Congress on Evolutionary Computation* (CEC04), Potland, Oregon. USA, pp. 1044-1051, 2004.
- [161] Jon Timmis and Camilla Edmonds, "A Comment on opt-AINet: An Immune Network Algorithm for Optimisation," *Proceedings, Part I. Genetic and Evolutionary Computation Conference (GECCO 2004)*, Seattle, WA, USA, pp. 308-317, 2004.
- [162] Jon Timmis and Mark J. Neal, A resource limited artificial immune system for data analysis *Knowledge Based Systems Journal: Special Issue*, vol. 14, no. 3-4, pp. 121-130, 2001.
- [163] Jon Timmis and Mark Neal, "Investigating the evolution and stability of a resource limited artificial immune system," Special Workshop on Artificial Immune Systems, Gentic and Evolutionay Computtion Conference (GECCO) 2000, Las Vegas, Nevada, U.S.A., pp. 40-41, 2000.
- [164] Jon Timmis, Mark Neal, and John E. Hunt, "Data analysis using artificial immune systems and cluster analysis and kohonen networks : Some comparisons," *Proceedings of International Conference Systems and Man and Cybernetics*, Tokyo, Japan, pp. 922-927, 1999.
- [165] Jon Timmis, Mark Neal, and John Hunt, An Artificial Immune System for Data

Analysis Biosystems, vol. 55, pp. 143-150, 2000.

- [166] Jon Timmis, Martyn Amos, Wolfgang Banzhaf, and Andy Tyrrell, "Going back to our roots": second generation biocomputing *International Journal of Unconventional Computing*, vol. Unpublished (Submitted 2005).
- [167] Jon Timmis and Thomas Knight. Chapter XI: Artificial immune systems: Using the immune system as inspiration for data mining. In: *Data Mining: A Heuristic Approach*, eds. Hussein A. Abbass, Ruhul A. Sarker, and Charles S. Newton. Group Idea Publishing, 2001.pp. 209-230.
- [168] Jon Timmis, Thomas Knight, Leandro N. de Castro, and Emma Hart. An Overview of Artificial Immune Systems. In: *Computation in Cells and Tissues: Perspectives and Tools for Thought*, Anonymous Springer, 2004.pp. 51-86.
- [169] Jonathan Ian Timmis, Artificial Immune Systems: A novel data analysis technique inspired by the immune network theory 2000. Department of Computer Science, University of Wales.
- [170] José Pacheco and José Félix Costa, Immune algorithms as formal probilistic algorithms *Evolutionary Computation Journal*, vol. Unpublished (Submitted 2004).
- [171] Juan Carlos Galeano, Angélica Veloza-Suan, and Fabio A. González, "A comparative analysis of artificial immune network models," *Proceedings of the 2005 conference on Genetic and evolutionary computation (GECCO'05)*, Washington DC, USA, pp. 361-368, 2005.
- [172] Julie Greensmith, New Frontiers For An Artificial Immune System 2003. University of Leeds.
- [173] Julie Greensmith, Jamie Twycross, and Uwe Aickelin, "Dendritic Cells for Anomaly Detection," *IEEE Congress on Evolutionary Computation (CEC06)*, Vancouver, BE, Canada, Unpublished (Accepted, 2006).
- [174] Julie Greensmith and Steve Cayzer, "An Artificial Immune System Approach to Semantic Document Classification," *Proceedings Artificial Immune Systems:* Second International Conference, ICARIS 2003, Edinburgh, UK, pp. 136-146, 2003.
- [175] Julie Greensmith and Uwe Aickelin, "Dendritic Cells for Real-Time Anomaly Detection," *Proceedings of the Workshop on Artificial Immune Systems and Immune System Modelling*, Bristol, UK, pp. 7-8, 2006.
- [176] Julie Greensmith, Uwe Aickelin, and Jamie Twycross, "Detecting Danger: Applying a Novel Immunological Concept to Intrusion Detection Systems'," *Proceedings 6th International Conference in Adaptive Computing in Design* and Manufacture, Bristol, UK, 2004.
- [177] Julie Greensmith, Uwe Aickelin, and Steve Cayzer, "Introducing dendritic cells as a novel immune inspired algorithm for anomaly detection," *Proceedings ICARIS-2005, 4th International Conference on Artificial Immune Systems,* Banff, Canada, pp. 153-167, 2005.
- [178] Jun Sun Yoo and Prabhat Hajela, Immune network simulations in multicriterion

design *Structural and Multidisciplinary Optimization*, vol. 18, pp. 85-94, Oct, 1999.

- [179] Jungwon Kim, A. Ong, and R. E. Overill, "Design of an artificial immune system as a novel anomaly detector for combating financial fraud in the retail sector," *The 2003 Congress on Evolutionary Computation, CEC '03*, pp. 405-412, 2003.
- [180] Jungwon Kim, Julie Greensmith, Jamie Twycross, and Uwe Aickelin, "Malicious Code Execution Detection and Response Immune System inspired by the Danger Theory," *Adaptive and Resilient Computing Security Workshop* (ARCS-05), Santa Fe, USA, 2005.
- [181] Jungwon Kim and Peter Bentley, "The Human Immune System and Network Intrusion Detection," 7th European Congress on Intelligent Techniques and Soft Computing (EUFIT '99), Aachen, Germany, 1999.
- [182] Jungwon Kim and Peter Bentley, Immune Memory and Gene Library Evolution in the Dynamic Clonal Selection Algorithm *Genetic Programming and Evolvable Machines*, vol. 5, pp. 361-391, Dec, 2004.
- [183] Jungwon Kim and Peter J. Bentley, "The Artificial Immune Model for Network Intrusion Detection," 7th European Congress on Intelligent Techniques and Soft Computing (EUFIT '99), Aachen, Germany, 1999.
- [184] Jungwon Kim and Peter J. Bentley, "Negative Selection and Niching by an Artificial Immune System for Network Intrusion Detection," A late-breaking paper, Genetic and Evolutionary Computation Conference (GECCO '99), Orlando, Florida, USA, 1999.
- [185] Jungwon Kim and Peter J. Bentley, "An Evaluation of Negative Selection in an Artificial Immune System for Network Intrusion Detection," *Proceedings of* the Genetic and Evolutionary Computation Conference (GECCO-2001), pp. 1330-1337, 2001.
- [186] Jungwon Kim and Peter J. Bentley, "Towards an artificial immune system for network intrusion detection: an investigation of clonal selection with a negative selection operator," *Proceedings of the 2001 Congress on Evolutionary Computation*, Seoul, South Korea, pp. 1244-1252, 2001.
- [187] Jungwon Kim and Peter J. Bentley, "Immune Memory in the Dynamic Clonal Selection Algorithm," *Proceedings of the First International Conference on Artificial Immune Systems (ICARIS)*, University of Kent, Canterbury, UK, pp. 57-65, 2002.
- [188] Jungwon Kim and Peter J. Bentley, "A Model of Gene Library Evolution in the Dynamic Clonal Selection Algorithm," *Proceedings of the First International Conference on Artificial Immune Systems (ICARIS)*, University of Kent at Canterbury, UK, pp. 175-182, 2002.
- [189] Jungwon Kim and Peter J. Bentley, "Towards an Artificial Immune System for Network Intrusion Detection: An Investigation of Dynamic Clonal Selection," *Congress on Evolutionary Computation (CEC-2002)*, Honolulu, USA, pp. 1015-1020, 2002.

- [190] Jungwong W. Kim, Integrating Artificial Immune Algorithms for Intrusion Detection 2002. Department of Computer Science, University College London.
- [191] Junichi Suzuki, Biologically-inspired Autonomous Adaptability in Communication Endsystem: An Approach Using an Artificial Immune Network 2001. Keio University.
- [192] Junichi Suzuki and Yoshikazu Yamamoto, "Autonomous decentralized control of distributed system configuration using an artificial immune system," *Proceedings of the International Workshop on Autonomous Decentralized System,* Chendu, China, 2000.
- [193] Junichi Suzuki and Yoshikazu Yamamoto, "Building an Artificial Immune Network for Decentralized Policy Negotiation in a Communication Endsystem: OpenWebServer/iNexus Study," Proceedings of The 4th World Multiconference on Systemics, Cybernetics and Informatics (SCI 2000), Orlando FL, USA, 2000.
- [194] Junichi Suzuki and Yoshikazu Yamamoto, "iNet: an extensible framework for simulating immune network," *IEEE International Conference on Systems, Man, and Cybernetics,* Nashville, TN, USA, pp. 119-124, 2000.
- [195] Junichi Suzuki and Yoshikazu Yamamoto, Biologically-inspired Autonomous Adaptability in a Communication Endsystem: An Approach Using an Artificial Immune Network *IEICE Transactions on Information and System*, vol. E84-D, pp. 1782-1789, Dec, 2001.
- [196] Justin Balthrop, Fernando Esponda, Stephanie Forrest, and Matthew Glickman, "Coverage and Generalization in an Artificial Immune System," *Proceedings* of the Genetic and Evolutionary Computation Conference (CEC02), pp. 3-10 , 2002.
- [197] K. M. Begnum and M. Burgess, "A scaled, immunological approach to anomaly countermeasures: combining pH with cfengine," *IFIP/IEEE Eighth International Symposium on Integrated Network Management*, pp. 31-42, 2003.
- [198] K. P. Anchor, J. B. Zydallis, G. H. Gunsch, and G. B. Lamont, "Extending thecomputer defense immune system: Network intrusion detection with a multiobjective evolutionary programming spproach," *First International Conference on Artificial Immune Systems (ICARIS 2002)*, University of Kent at Canterbury, pp. 12–21, 2002.
- [199] K. P. Anchor, P. D. Williams, G. H. Gunsch, and G. B. Lamont, "The computer defense immune system: current and future research in intrusion detection," *Proceedings of the 2002 Congress on Evolutionary Computation (CEC '02)*, Honolulu, HI, USA, pp. 1027-1032, 2002.
- [200] Kathy Jean Matthews, Immunotronics: Self-repairing finite state machines 2003. University of West England.
- [201] Kazuyuki Mori, Makoto Tsukiyama, and Toyoo Fukuda, Immune algorithm with searching diversity and its application to resource allocation problem *The Transactions of the Institute of Electrical Engineers of Japan*, vol. 113-C, pp. 872-878, 1993.

- [202] L. N. de Castro, Immune Cognition, Micro-evolution, and a Personal Account on Immune Engineering S.E.E.D. Journal (Semiotics, Evolution, Energy, and Development), vol. 3, pp. 134-155, 2003.
- [203] Leandro N. de Castro, "The immune response of an artificial immune network (aiNet)," 8-12 Dec. 2003, pp. 146-153, 2003.
- [204] Leandro N. de Castro and Fernando J. Von Zuben, "The Clonal Selection Algorithm with Engineering Applications," *Proceedings of the Genetic and Evolutionary Computation Conference (GECCO '00), Workshop on Artificial Immune Systems and Their Applications*, Las Vegas, Nevada, USA, pp. 36-37, 2000.
- [205] Leandro N. de Castro and Fernando J. Von Zuben, "An evolutionary immune network for data clustering," *Proceedings Sixth Brazilian Symposium on Neural Networks*, Rio de Janeiro, RJ, Brazil, pp. 84-89, 2000.
- [206] Leandro N. de Castro and Fernando J. Von Zuben. Chapter XII: aiNet: An Artificial Immune Network for Data Analysis. In: *Data Mining: A Heuristic Approach*, eds. H. A. Abbass, R. A. Sarker, and C. S. Newton. USA: Idea Group Publishing, 2001.pp. 231-259.
- [207] Leandro N. de Castro and Fernando J. Von Zuben, Immune and Neural Network Models: Theoretical and Emperical Comparisons International journale of computational ingelligence and applications (IJCIA), vol. 1, pp. 239-257, 2001.
- [208] Leandro N. de Castro and Fernando J. Von Zuben, Learning and optimization using the clonal selection principle *IEEE Transactions on Evolutionary Computation*, vol. 6, pp. 239-251, Jun, 2002.
- [209] Leandro N. de Castro and Fernando José Von Zuben, "Artificial Immune Systems -Part I: Basic Theory and Applications," Department of Computer Engineering and Industrial Automation, School of Electrical and Computer Engineering, State University of Campinas, Brazil, TR DCA 01/99, Dec 1999.
- [210] Leandro N. de Castro and Fernando José Von Zuben, "Artificial Immune Systems: Part II: A Survey of Applications," Department of Computer Engineering and Industrial Automation, School of Electrical and Computer Engineering, State University of Campinas, Brazil, DCA-RT 02/00, Feb 2000.
- [211] Leandro N. de Castro and Jon Timmis, "An artificial immune network for multimodal function optimization," *Proceedings of the 2002 Congress on Evolutionary Computation (CEC '02)*, Honolulu, HI, USA, pp. 699-704, 2002.
- [212] Leandro N. de Castro and Jon Timmis. *Artificial Immune Systems: A new computational intelligence approach*, Great Britain: Springer-Verlag, 2002.
- [213] Leandro N. de Castro and Jon Timmis. Artificial Immune Systems: A Novel Paradigm to Pattern Recognition. In: Artificial Neural Networks in Pattern Recognition, eds. J. M. Corchado, L. Alonso, and C. Fyfe. UK: University of Paisley, 2002.pp. 67-84.
- [214] Leandro N. de Castro and Jon Timmis, "Hierarchy and Convergence of Immune

Networks: Basic Ideas and Preliminary Results," *Proceedings of the 1st International Conference on Artificial Immune Systems (ICARIS'02),* University of Kent at Canterbury, UK, pp. 231-240, 2002.

- [215] Leandro N. de Castro and Jon Timmis, Artificial immune systems as a novel soft computing paradigm Soft Computing - A Fusion of Foundations, Methodologies and Applications, vol. 7, pp. 526-544, Aug, 2003.
- [216] Leandro Nunes de Castro, Fernando J. Von Zuben, and Getúlio A. de Deus Jr., The Construction of a Boolean Competitive Neural Network Using Ideas From Immunology *Neurocomputing*, vol. 50, pp. 51-85, Jan, 2003.
- [217] Lee A. Segel, "The immune system as a prototype of autonomous decentralized systems," *IEEE International Conference on Systems, Man, and Cybernetics,* 1997. 'Computational Cybernetics and Simulation, Orlando, FL, USA, pp. 375-385, 1997.
- [218] Lee A. Segel, Diffuse feedback from diffuse information in complex systems *Complexity*, vol. 5, pp. 39-46, 2000.
- [219] Lee A. Segel, Controlling the immune system: Diffuse feedback via a diffuse informational network Complexity in Biological Information Processing, Novartis Foundation Symposium, vol. 239, pp. 31-51, 2001.
- [220] Lee A. Segel. Diffuse feedback from a diffuse informational network: in the immune system and other distributed autonomous systems. In: *Design Principles for the Immune System and Other Distributed Autonomous Systems*, ed. Lee A. Segel, I. R. C. New York: Oxford University Press, 2001.pp. 203-226.
- [221] Lee A. Segel, How can perception of context improve the immune response? *Autoimmunity and Emerging Disease*, vol. pp. 169-191, 2001.
- [222] Lee A. Segel, How does the immune system see to it that it is doing a good? *Graft*, vol. 6, pp. 15-18, 2001.
- [223] Lee A. Segel, Some Spatio-temporal models in immunology *International Journal* of *Bifurcation and Chaos*, vol. 12, pp. 2343-2347, 2002.
- [224] Lee A. Segel and Irun R. Cohen. *Design Principles for the Immune System and Other Distributed Autonomous Systems*, USA: Oxford University Press, 2001.
- [225] Lee A. Segel and Ruth Lev Bar-Or. Immunology Viewed as the study of an Autonomous Decentralized System. In: Artificial Immune Systems and Their Applications, ed. D. Dasgupta. Germany: Springer, 1999.pp. 65-88.
- [226] Lee A. Segel and Ruth Lev Bar-Or, On the role of feedback in promoting conflicting goals of the adaptive immune system *The Journal of Immunology*, vol. pp. 1342-1349, 1999.
- [227] Liu Shulin, Zhang Jiazhong, Shi Wengang, and Huang Wenhu , "Negative-selection algorithm based approach for fault diagnosis of rotary machinery," *Proceedings of the 2002 American Control Conference*, pp. 3955-3960, 2002.

- [228] Lois Boggess and Janna S. Hamaker, "The Effect of Irrelevant Features on AIRS, an Artificial Immune-Based Classifier," *Intelligent Engineering Systems through Artificial Neural Networks (ANNIE)*, pp. 219-224, 2003.
- [229] M. Araujo, J. Aguilar, and H. Aponte, "Fault detection system in gas lift well based on artificial immune system," *Proceedings of the International Joint Conference on Neural Networks*, pp. 1673-1677, 2003.
- [230] M. Ayara, Jon Timmis, R. de Lemos, Leandro N. de Castro, and R. Duncan, "Negative selection: How to generate detectors," *1st International Conference on Artificial Immune Systems*, University of Kent at Canterbury, pp. 89-98, 2002.
- [231] M. Kayama, Y. Sugita, Y. Morooka, and S. Fukuoka, "Distributed Diagnosis System Combining the Immune Network and Learning Vector Quantization," *Proceedings of the 1995 IEEE IECON 21st International Conference on Industrial Electronics, Control, and Instrumentation (IECON'95),* Orlando, FL, USA, pp. 1531-1536, 1995.
- [232] Marc Ebner, Hans-Georg Breunig, and Jürgen Albert, "On The Use Of Negative Selection In An Artificial Immune System," *Proceedings of the Genetic and Evolutionary Computation Conference (GECCO 2002)*, New York, NY, USA, pp. 957-964, 2002.
- [233] Mario Villalobos-Arias, Carlos A. Coello Coello, and Onésimo Hernández-Lerma, "Convergence Analysis of a Multiobjective Artificial Immune System Algorithm," *Proceedings Artificial Immune Systems: Third International Conference, ICARIS 2004*, Catania, Sicily, Italy, pp. 226-235, 2004.
- [234] Mark Burgess, A site configuration engine USENIX Computing systems, vol. 8, 1995.
- [235] Mark Burgess, "Computer Immunology," *Proceedings of the 12th USENIX conference on System administration*, pp. 283-298, 1998.
- [236] Mark Burgess, "Evaluating cfengine's Immunity Model of System Maintenance," *Proceedings of the 2nd SANE system administration conference (USENIX/NLUUG)*, 2000.
- [237] Mark Burgess, Probabilistic anomaly detection in distributed computer networks *Science of Computer Programming*, vol. 60, pp. 1-26, 2006.
- [238] Mark Burgess and R. Ralston, Distributed resource administration using cfengine *Software-Practice and Experience*, vol. 27, 1997.
- [239] Mark Neal, "An artificial immune system for continuous analysis of time-varying data," *Proceedings of the 1st International Conference on Artificial Immune Systems (ICARIS)*, University of Kent at Canterbury, pp. 76-85, 2002.
- [240] Mark Neal, "Meta-stable Memory in an Artificial Immune Network," Proceedings Artificial Immune Systems, Second International Conference, ICARIS 2003, Edinburgh, UK, pp. 168-180, 2003.
- [241] Mark Neal and Jon Timmis. Chapter XIV: Once More Unto the Breach: Towards Artificial Homeostasis? In: *Recent Developments in Biologically Inspired*

Computing, eds. Leandro N. de Castro and F. J. Von Zuben. USA: Idea Group, 2005.pp. 340-365.

- [242] N. Toma, S. Endo, K Yamada, and H. Miyagi, "The immune distributed competitive problem solver with majorhistocompatibility complex and immune network," *IEEE International Conference on Systems, Man, and Cybernetics, Nashville,* TN, USA, pp. 1865-1870, 2000.
- [243] Nareli Cruz Cortés and Carlos A. Coello Coello, "Multiobjective Optimization Using Ideas from the Clonal Selection Principle," *Proceedings, Part I Genetic and Evolutionary Computation Conference (GECCO 2003)*, Chicago, IL, USA, pp. 158-170, 2003.
- [244] Nareli Cruz Cortés, Sistema Inmune Artificial para Solucionar Problemas de Optimización (Artificial immune system to solve problems of optimization) 2004. The Evolutionary Computation Group at CINVESTAV-IPN (EVOCINV).
- [245] Niloy Ganguly, Geoff Canright, and Andreas Deutsch, "Design of an efficient search algorithm for p2p networks using concepts from natural immune systems," *Proceedings of the 8th International Conference on Parallel Problem Solving from Nature (PPSN VIII)*, Birmingham, UK, 2004.
- [246] Olfa Nasraoui, Dipankar Dasgupta, and Fabio A. González, "A Novel Artificial Immune System Approach to Robust Data Mining," *Late Breaking papers at the Genetic and Evolutionary Computation Conference (GECCO-2002)*, New York, USA, pp. 356-363, 2002.
- [247] Olfa Nasraoui, Dipankar Dasgupta, and Fabio Gonzalez, "The Promise and Challenges of Artificial Immune System Based Web Usage Mining: Preliminary Results," *SIAM Workshop on Web Analytics*, Arlington, VA, USA, pp. 29-39, 2002.
- [248] Olfa Nasraoui, Fabio Gonzalez, Cesar Cardona, Carlos Rojas, and Dipankar Dasgupta, "A Scalable Artificial Immune System Model for Dynamic Unsupervised Learning," *Proceedings, Part I Genetic and Evolutionary Computation Conference (GECCO)*, Chicago, IL, USA, pp. 219-230, 2003.
- [249] Olfa Nasraoui, Fabio Gonzalez, and Dipankar Dasgupta, "The fuzzy artificial immune system: motivations, basic concepts, and application to clustering and Web profiling," *Proceedings of the 2002 IEEE International Conference on Fuzzy Systems (FUZZ-IEEE'02.)*, Honolulu, HI, USA, pp. 711-716, 2002.
- [250] P. Bull, A. Knowles, G. Tedesco, and A. Hone, "Diophantine Benchmarks for the B-Cell Algorithm," *Proceedings 5th International Conference on Artificial Immune Systems (ICARIS 2006)*, Instituto Gulbenkian de Cięncia, Oeiras, Portugal, 2006.
- [251] P. K. Harmer, P. D. Williams, G. H. Gunsch, and G. B. Lamont, An artificial immune system architecture for computer security applications *IEEE Transactions on Evolutionary Computation*, vol. 6, pp. 252-280, Jun, 2002.
- [252] Patrik D'haeseleer, "An immunological approach to change detection: theoretical results," *Proceedings of the 9th IEEE Computer Security Foundations Workshop*, Kenmare, Ireland, pp. 18-26, 1996.

- [253] Patrik D'haeseleer, S. Forrest, and P. Helman, "An immunological approach to change detection: algorithms, analysis and implications," *IEEE Symposium* on Security and Privacy, Oakland, CA, USA, pp. 110-119, 1996.
- [254] Paul D. Williams, Kevin P. Anchor, John L. Bebo, Gregg H. Gunsch, and Gary D. Lamont, "CDIS: Towards a Computer Immune System for Detecting Network Intrusions," *Proceedings 4th International Symposium: Recent Advances in Intrusion Detection (RAID)*, Davis, CA, USA, pp. 117-133, 2001.
- [255] Paul K. Harmer, A Distributed Agent Architecture for a Computer Virus Immune System 2000. Air Force Institute of Technology, Air University.
- [256] Paul S. Andrews and Jon Timmis, "Inspiration for the Next Generation of Artificial Immune Systems," *Proceedings Artificial Immune Systems: 4th International Conference, ICARIS 2005*, Banff, Alberta, Canada, pp. 126-138, 2005.
- [257] Peter J. Bentley, "Why Biologists and Computer Scientists Should Work Together," Selected Papers from the 5th European Conference on Artificial Evolution, pp. 3-18, 2001.
- [258] Peter J. Bentley, Julie Greensmith, and Supiya Ujjin, "Two Ways to Grow Tissue for Artificial Immune Systems," *Proceedings Artificial Immune Systems: 4th International Conference, ICARIS 2005, Banff, Alberta, Canada, pp. 139, 2005.*
- [259] Philipp H. Mohr, Nick Ryan, and Jon Timmis, "Exploiting Immunological Properties for Ubiquitous Computing Systems," *Proceedings Artificial Immune Systems: Third International Conference, ICARIS 2004*, Catania, Sicily, Italy, pp. 277-289, 2004.
- [260] Prabhat Hajela and J. Lee, Constrained genetic search via schema adaptation: An immune network solution *Structural and Multidisciplinary Optimization*, vol. 12, pp. 11-15, Aug, 1996.
- [261] Prabhat Hajela and Jun Sun Yoo. Chapter Thirteen: Immune Network Modelling in Design Optimization. In: *New Ideas in Optimization*, eds. D. Corne, M. Dorigo, and F. Glover. England: McGraw-Hill, 1999.pp. 203-216.
- [262] R O Canham and A M Tyrrell, "A Multilayered Immune System for Hardware Fault Tolerance within an Embryonic Array," *Proceedings of the 1st International Conference on Artificial Immune Systems (ICARIS)*, University of Kent at Canterbury, UK, pp. 3-11, 2002.
- [263] R. O. Canham and A. M. Tyrrell, A Hardware Artificial Immune System and Embryonic Array for Fault Tolerant Systems *Journal of Genetic Programming and Evolvable Machine*, vol. 4, pp. 359-382, 2003.
- [264] Richard G. Weinand, Somatic mutation, affinity maturation and the antibody repertoire: a computer model *Journal of Theoretical Biology*, vol. 143, pp. 343-382, 1990.
- [265] Rob J. de Boer and Alan S. Perelson, How diverse should the immune system be? Proceedings Biological sciences / The Royal Society London, vol. 252, pp. 171-175, 1993.

- [266] Robert E. Smith, Stephanie Forrest, and Alan S. Perelson, "Population Diversity in an Immune System Model: Implications for Genetic Search," *Proceedings of the Second Workshop on Foundations of Genetic Algorithms (FOGA)*, Vail, Colorado, USA, pp. 153-165, 1993.
- [267] Robert E. Smith, Stephanie Forrest, and Alan S. Perelson, Searching for diverse, cooperative populations with genetic algorithms *Evolutionary Computation*, vol. 1, pp. 127-149, 1993.
- [268] Roberto Michelan and Fernando J. Von Zuben, "Decentralized control system for autonomous navigation based on an evolved artificial immune network," *Proceedings of the 2002 Congress on Evolutionary Computation, (CEC '02),* Honolulu, HI, USA, pp. 1021-1026, 2002.
- [269] Ron Hightower, Stephanie Forrest, and Alan S. Perelson, "The Evolution of Secondary Organization in Immune System Gene Libraries," *Proceedings of the Second European Conference on Artificial Life (ECAL '93)*, Brussels, Belgium, 1993.
- [270] Ron Hightower, Stephanie Forrest, and Alan S. Perelson, "The Baldwin Effect in the Immune System: Learning by Somatic Hypermutation," *Adaptive Individuals* in Evolving Populations: Models and Algorithms, pp. 159-167, 1995.
- [271] Ron R. Hightower, Computational Aspects of Antibody Gene Families 1996. University of New Mexico.
- [272] Ron R. Hightower, Stephanie Forrest, and Alan S. Perelson, "The Evolution of Emergent Organization in Immune System Gene Libraries," *Proceedings of* the 6th International Conference on Genetic Algorithms, pp. 344-350, 1995.
- [273] Scott M. Thayer and Surya P. N. Singh, "Development of an immunology-based multi-robot coordination algorithm for exploration and mapping domains," *IEEE/RSJ International Conference on Intelligent Robots and System*, pp. 2735-2739, 2002.
- [274] Shantanu Singh, "Anomaly detection using negative selection based on the rcontiguous matching rule," *Proceedings of the 1st International Conference* on Artificial Immune Systems (ICARIS 2002), University of Kent, UK, pp. 99-106, 2002.
- [275] Simon M. Garrett, "A Paratope Is Not an Epitope: Implications for Immune Network Models and Clonal Selection," *Proceedings Artificial Immune Systems: Second International Conference, ICARIS 2003,* Edinburgh, UK, pp. 217-228, 2003.
- [276] Simon M. Garrett, "Parameter-free, adaptive clonal selection," Congress on Evolutionary Computing (CEC 2004), Portland Oregon, USA, pp. 1052-1058, 2004.
- [277] Simon M. Garrett, How do we evaluate artificial immune systems? *Evolutionary Computation*, vol. 13, pp. 145-177, 2005.
- [278] Slavisa Sarafijanovic and Jean-Yves Le Boudec, "An Artificial Immune System for Misbehavior Detection in Mobile Ad-Hoc Networks with Virtual Thymus, Clustering, Danger Signal, and Memory Detectors," *Proceedings Artificial*

Immune Systems: Third International Conference, ICARIS 2004, Catania, Sicily, Italy, pp. 342-356, 2004.

- [279] Slavisa Sarafijanovic and Jean-Yves Le Boudec, An artificial immune system approach with secondary response for misbehavior detection in mobile ad hoc networks *IEEE Transactions on Neural Networks*, vol. 16, pp. 1076-1087, Sep, 2005.
- [280] Slawomir T. Wierzchon, Discriminative power of the receptors activated by kcontiguous bits rule *Journal of Computer Science and Technology*, vol. 1, pp. 1-13, 2000.
- [281] Slawomir T. Wierzchon, "Generating Optimal Repertoire of Antibody Strings in an Artificial Immune System," *Proceedings of the IIS'2000 Symposium on Intelligent Information Systems*, pp. 119-133, 2000.
- [282] Slawomir T. Wierzchon, "Deriving a concise description of non-self patterns in an artificial immune system," *New learning paradigms in soft computing*, pp. 438-464, 2002.
- [283] Slawomir T. Wierzchon and U. Kuzelewska., "Stable Clusters Formation in an Artificial Immune System," *First International Conference on Artificial Immune Systems*, University of Kent at Canterbury, pp. 68-75, 2002.
- [284] Spyros Xanthakis, Constantinos Karapoulios, Régis Pajot, and Ahmed Rozz,
 "Immune System and Fault-Tolerant Computing," *Selected Papers from the European conference on Artificial Evolution*, pp. 181-197, 1995.
- [285] Srividhya Sathyanath and Ferat Sahin, "An AIS approach to a color image classification problem in a real-time industrial application," *IEEE International Conference on Systems, Man, and Cybernetics,* Tucson, AZ, USA, pp. 2285-2290, 2001.
- [286] Srividhya Sathyanath and Ferat Sahin, "AISIMAM An Artificial Immune System Based Intelligent Multi-Agent Model and its Application to a Mine Detection Problem," *ICARIS 2002: 1st International Conference on Artificial Immune Systems,* University of Kent, pp. 22-31, 2002.
- [287] Stephanie Forrest, A. Somayaji, and D. H. Ackley, "Building diverse computer systems," *The Sixth Workshop on Hot Topics in Operating Systems*, Cape Cod, MA, USA, pp. 67-72, 1997.
- [288] Stephanie Forrest and Alan S. Perelson, "Genetic Algorithms and the Immune System," Proceedings of the 1st Workshop on Parallel Problem Solving from Nature, Dortmund, Germany, pp. 320-325, 1991.
- [289] Stephanie Forrest, Alan S. Perelson, Lawrence Allen, and Rajesh Cherukuri, "Self-Nonself Discrimination in a Computer," *Proceedings of the 1992 IEEE Symposium on Security and Privacy*, pp. 202-212, 1994.
- [290] Stephanie Forrest, Brenda Javornik, Robert E. Smith, and Alan S. Perelson, Using Genetic Algorithms to Explore Pattern Recognition in the Immune System *Evolutionary Computation*, vol. 1, pp. 191-211, 1993.
- [291] Stephanie Forrest, Justin Balthrop, Matthew Glickman, and David Ackley.

Computation in the Wild. In: *The Internet as a Large-Scale Complex System*, eds. Kihong Park and Walter Willinger. USA: Oxford University Press, 2005.

- [292] Stephanie Forrest and S. A. Hofmeyr. Immunology as Information Processing. In: Design Principles for the Immune System and Other Distributed Autonomous Systems, ed. Lee A. Segel, I. R. C. New York: Oxford University Press, 2001.pp. 361-388.
- [293] Stephanie Forrest, S. A. Hofmeyr, A. Somayaji, and T. A. Longstaff, "A Sense of Self for Unix Processes," *Proceedings of the 1996 IEEE Symposium on Security and Privacy*, Oakland, CA, USA, pp. 120-128, 1996.
- [294] Stephanie Forrest and Steven A. Hofmeyr, "John Holland's invisible hand: An artificial immune system," *Presented at the Festschrift held in honor of John Holland*, University of Michigan, USA, 1999.
- [295] Stephanie Forrest, Steven A. Hofmeyr, and Anil Somayaji, Computer immunology *Communications of the ACM*, vol. 40, pp. 88-96, Oct, 1997.
- [296] Steve Cayzer and Uwe Aickelin, "On the Effects of Idiotypic Interactions for Recomendation Communities in Artificial Immune Systems," *Proceedings of the 1st International Conference on Artificial Immune Systems (ICARIS)*, University of Kent at Canterbury, pp. 154-160, 2002.
- [297] Steve Cayzer and Uwe Aickelin, "A recommender system based on the immune network," *Proceedings of the 2002 Congress on Evolutionary Computation* (CEC '02), Honolulu, HI, USA, pp. 807-812, 2002.
- [298] Steve Cayzer and Uwe Aickelin, A Recommender System based on Idiotypic Artificial Immune Networks *Journal of Mathematical Modelling and Algorithms*, vol. 4, pp. 181-198, 2005.
- [299] Steven A. Hofmeyr and Stephanie Forrest, "Immunizing Computer Networks: Getting All the Machines in Your Network to Fight the Hacker Disease," *Proceedings 1999 IEEE Symposium on Security and Privacy*, 1999.
- [300] Steven A. Hofmeyr and Stephanie Forrest, Architecture for an Artificial Immune System *Evolutionary Computation*, vol. 8, pp. 443-473, Dec 1, 2000.
- [301] Steven A. Hofmeyr, Stephanie Forrest, and Anil Somayaji, Intrusion Detection using Sequences of System Calls *Journal of Computer Security*, vol. 6, pp. 151-180, 1998.
- [302] Steven Andrew Hofmeyr, An Immunological Model of Distributed Detection and its Application to Computer Security 1999. Department of Computer Sciences, University of New Mexico.
- [303] Steven Hofmeyr and Stephanie Forrest, "Immunity by Design: An Artificial Immune System," *Proceedings of the Genetic and Evolutionary Computation Conference (GECCO)*, San Francisco, CA, USA, pp. 1289-1296, 1999.
- [304] Steven Hofmeyr, Stephanie Forrest, and Patrik D'haeseleer, "An Immunological Approach to Distributed Network Intrusion Detection," *First International Workshop on the Recent Advances in Intrusion Detection (RAID'98*

Symposium), Louvain-la-Neuve, Belgium, 1998.

- [305] Surya P. N. Singh and Scott M. Thayer, "Immunology Directed Methods for Distributed Robotics: A Novel, Immunity-Based Architecture for Robust Control & Coordination," *Proceedings of SPIE: Mobile Robots XVI*, 2001.
- [306] Surya P. N. Singh and Scott M. Thayer, Kilorobot Search and Rescue Using an Immunologically Inspired Approach *Distributed Autonomous Robotic Systems*, vol. 5, Jun, 2002.
- [307] Surya P. N. Singh, Scott M. Thayer, and Wesley P. Thayer, "A foundation for kilorobotic exploration," *Proceedings of the 2002 Congress on Evolutionary Computation, CEC '02.* Honolulu, HI, USA, pp. 1033-1038, 2002.
- [308] Susan Stepney, John A. Clark, Colin G. Johnson, Derek Partridge, and Robert E. Smith, "Artificial immune systems and the grand challenge for non-classical computation," *Proceedings of the 2003 International Conference on Artificial Immune Systems*, pp. 204-216, 2003.
- [309] Susan Stepney, Robert E. Smith, Jonathan Timmis, and Andy M. Tyrrell, "Towards a Conceptual Framework for Artificial Immune Systems," *Proceedings Artificial Immune Systems: Third International Conference, ICARIS 2004*, Catania, Sicily, Italy, pp. 53-64, 2004.
- [310] Susan Stepney, Robert E. Smith, Jonathan Timmis, Andy M. Tyrrell, Mark J. Neal, and Andrew N. W. Hone, Conceptual Frameworks for Artificial Immune Systems *International Journal of Unconventional Computing*, vol. 1, pp. 315-338, Jul, 2005.
- [311] Terri Oda, A Spam-Detecting Artificial Immune System 2005. Ottawa-Charleton Institute for Computer Science, School of Computer Science, Carleton University.
- [312] Terri Oda and Tony White, "Developing an immunity to spam," *Proceedings, Part I Genetic and Evolutionary Computation Conference GECCO 2003*, pp. 231-242, 2003.
- [313] Terri Oda and Tony White, "Increasing the accuracy of a spam-detecting artificial immune system," *The 2003 Congress on Evolutionary Computation, CEC* '03, pp. 390-396, 2003.
- [314] Terri Oda and Tony White, "Immunity from Spam: An Analysis of an Artificial Immune System for Junk Email Detection," *Proceedings Artificial Immune Systems: 4th International Conference, ICARIS 2005*, Banff, Alberta, Canada, pp. 276, 2005.
- [315] Thomas Knight and Jon Timmis, "AINE: An Immunological Approach to Data Mining," *First IEEE International Conference on Data Mining (ICDM'01)*, San Jose, CA. USA, pp. 297-304, 2001.
- [316] Thomas Knight and Jon Timmis, "Assessing the performance of the resource limited artificial immune system AINE," Computing Laboratory, University of Kent at Canterbury, Canterbury, Kent, UK, 3-01, May 2001.
- [317] Thomas Knight and Jon Timmis, "A Multi-Layered Immune Inspired Approach to

Data Mining," *Proceedings of the 4th International Conference on Recent Advances in Soft Computing*, Nottingham, UK, pp. 266-271, 2002.

- [318] Thomas Knight and Jon Timmis, "A Multi-layered Immune Inspired Machine Learning Algorithm," *Applications and Science in Soft Computing*, pp. 195-202, Dec. 2003.
- [319] Thomas Stibor, On the Appropriateness of Negative Selection for Anomaly Detection and Network Intrusion Detection 2006. Darmstadt University of Technology.
- [320] Thomas Stibor, J. Timmis, and C. Eckert, "On the appropriateness of negative selection defined over Hamming shape-space as a network intrusion detection system," *The 2005 IEEE Congress on Evolutionary Computation, (CEC 2005)*, pp. 995-1002, 2005.
- [321] Thomas Stibor, Jonathan Timmis, and Claudia Eckert, "A Comparative Study of Real-Valued Negative Selection to Statistical Anomaly Detection Techniques," *Proceedings Artificial Immune Systems: 4th International Conference, ICARIS 2005*, Banff, Alberta, Canada, pp. 262, 2005.
- [322] Thomas Stibor, Kpatscha M. Bayarou, and Claudia Eckert, "An Investigation of R-Chunk Detector Generation on Higher Alphabets," *Proceedings, Part I Genetic and Evolutionary Computation Conference (GECCO 2004)*, Seattle, WA, USA, pp. 299-307, 2004.
- [323] Thomas Stibor, P. Mohr, J. Timmis, and C. Eckert, "Is negative selection appropriate for anomaly detection?," *Genetic and Evolutionary Computation Conference* (*GECCO*), Washington DC, USA, pp. 321-328, 2005.
- [324] Tom Morrison, Similarity Measure Building for Website Recommendation within an Artificial Immune System 2003. School of Computer Science, University of Nottingham.
- [325] Tom Morrison and Uwe Aickelin, "An Artificial Immune System as a Recommender System for Web Sites," *Proceedings 1st International Conference on Artificial Immune Systems (ICARIS - 2002),* University of Kent at Canterbury, UK, pp. 161-169, 2002.
- [326] Toyoo Fukuda, Kazuyuki Mori, and Makoto Tsukiyama, Immune networks using genetic algorithm for adaptive production scheduling 15th IFAC World Congress, vol. 3, pp. 57-60, 1993.
- [327] Toyoo Fukuda, Kazuyuki Mori, and Makoto Tsukiyama. Chapter 11: Parallel Search for Multi-Modal Function Optimization with Diversity and Learning of Immune Algorithm. In: Artificial Immune Systems and Their Applications, ed. Dipankar Dasgupta. Springer-Verlag, 1999.pp. 210-220.
- [328] Uwe Aickelin, Julie Greensmith, and Jamie Twycross, "Immune System Approaches to Intrusion Detection A Review," *Proceedings of the Third International Conference on Artificial Immune Systems (ICARIS-04)*, pp. 316-329, 2004.
- [329] Uwe Aickelin, Peter Bentley, Steve Cayzer, Jungwon Kim, and Julie McLeod, "Danger Theory: The Link between AIS and IDS?," *Proceedings of the 2nd International Conference on Artificial Immune Systems (ICARIS-2003),*

Edinburgh, UK, pp. 147-155, 2003.

- [330] Uwe Aickelin and Q. Chen, "Movie Recommendation Systems Using An Artificial Immune System," Proceedings of 6th International Conference in Adaptive Computing in Design and Manufacture, Bristol, UK, pp. http://www.asap.cs.nott.ac.uk/publications/conf-papers.shtml, 2004.
- [331] Uwe Aickelin and Q. Chen, "On Affinity Measures for Artificial Immune System Movie Recommenders," *Proceedings RASC-2004, The 5th International Conference on: Recent Advances in Soft Computing*, Nottingham, UK, 2004.
- [332] Uwe Aickelin and Steve Cayzer, "The Danger Theory and Its Application to Artificial Immune Systems," *Proceedings 1st International Conference on Artificial Immune Systems (ICARIS - 2002),* Canterbury, pp. 141-148, 2002.
- [333] V. Cutello, G. Narzisi, G. Nicosia, and M. Pavone, "An Immunological Algorithm for Global Numerical Optimization," 7th International Conference on Artificial Evolution (EA), Lille, France, pp. 284-295, 2005.
- [334] V. Cutello, G. Nicosia, M. Pavone, and G. Narzisi, "Real coded clonal selection algorithm for unconstrained global optimization using a hybrid inversely proportional hypermutation operator," *Proceedings of the 2006 ACM symposium on Applied computing*, Dijon, France, pp. 950-954, 2006.
- [335] Vincenzo Cutello, G. Morelli, Giuseppe Nicosia, and Mario Pavone, "Immune Algorithms with Aging Operators for the String Folding Problem and the Protein Folding Problem," *Proceedings Evolutionary Computation in Combinatorial Optimization: 5th European Conference (EvoCOP 2005)*, Lausanne, Switzerland, pp. 80-90, 2005.
- [336] Vincenzo Cutello, G. Narzisi, Giuseppe Nicosia, Mario Pavone, and G. Sorace, "How to Escape Traps using Clonal Selection Algorithms," *The First International Conference on Informatics in Control, Automation and Robotics, ICINCO* 2004, Setubal, Portugal, pp. 322-326, 2004.
- [337] Vincenzo Cutello, Giuseppe Narzisi, Giuseppe Nicosia, and Mario Pavone, "Clonal Selection Algorithms: A Comparative Case Study using Effective Mutation Potentials," *Proceedings Artificial Immune Systems: 4th International Conference, ICARIS 2005*, Banff, Alberta, Canada, pp. 13-28, 2005.
- [338] Vincenzo Cutello and Giuseppe Nicosia, "An Immunological Approach to Combinatorial Optimization Problems," *Proceedings of the 8th Ibero-American Conference on AI: Advances in Artificial Intelligence*, Seville, Spain, pp. 361-370, 2002.
- [339] Vincenzo Cutello and Giuseppe Nicosia, "Multiple learning using immune algorithms," Proceedings of 4th International Conference on Recent Advances in Soft Computing, RASC 2002, Nottingham, UK, pp. 102-107, 2002.
- [340] Vincenzo Cutello and Giuseppe Nicosia. Chapter VI. The Clonal Selection Principle for In Silico and In Vivo Computing. In: *Recent Developments in Biologically Inspired Computing*, eds. Leandro Nunes de Castro and Fernando J. Von Zuben. Hershey, London, Melbourne, Singapore: Idea Group Publishing, 2005.pp. 104-146.

- [341] Vincenzo Cutello, Giuseppe Nicosia, and Mario Pavone, "A Hybrid Immune Algorithm with Information Gain for the Graph Coloring Problem," *Proceedings, Part I: Genetic and Evolutionary Computation Conference* (GECCO 2003), Chicago, IL, USA, pp. 171-182, 2003.
- [342] Vincenzo Cutello, Giuseppe Nicosia, and Mario Pavone, "Exploring the Capability of Immune Algorithms: A Characterization of Hypermutation Operators," *Proceedings Artificial Immune Systems: Third International Conference, ICARIS 2004,* Catania, Sicily, Italy, pp. 263-276, 2004.
- [343] Vincenzo Cutello, Giuseppe Nicosia, and Mario Pavone, "An immune algorithm with hyper-macromutations for the Dill's 2D hydrophobic-hydrophilic model," *Congress on Evolutionary Computation, (CEC2004),* pp. 1074-1080, 2004.
- [344] X. Wang, X. Z. Gao, and S. J. Ovaska, "Artificial immune optimization methods and applications a survey," *IEEE International Conference on Systems, Man and Cybernetics*, pp. 3415-3420, 2004.
- [345] X. Zhang, G. Dragffy, A.G. Pipe, and Q.M. Zhu, "Artificial Innate Immune System: An Instant Defence Layer of Embryonics," *Proceedings Artificial Immune Systems: Third International Conference, ICARIS 2004*, Catania, Sicily, Italy, pp. 302-315, 2004.
- [346] Yoshiteru Ishida, "Fully distributed diagnosis by PDP learning algorithm: towards immune network PDP model," *IJCNN International Joint Conference on Neural Networks*, San Diego, CA, USA, pp. 777-782, 1990.
- [347] Yoshiteru Ishida, Immune network model and its applications to process diagnosis *Systems and Computers in Japan*, vol. 24, pp. 38-46, 1993.
- [348] Yoshiteru Ishida, An Immune Network Approach to Sensor-based Diagnosis by Selforganization *Complex Systems*, vol. 10, pp. 73-90, 1996.
- [349] Yoshiteru Ishida, "The Immune System as a Self-Identification Process: A Survey and a Proposal," *Proceeding of ICMAS International Workshop on Immunity-Based Systems (IMBS96)*, Kyoto, Japan, pp. 2-12, 1996.
- [350] Yoshiteru Ishida, "Active Diagnosis Self-Organization: An Approach The Immune Network Metaphor," *International joint conference on artificial intelligence* (*IJCAI*), pp. 1084-1091, 1997.
- [351] Yoshiteru Ishida, "The immune system as a prototype of autonomous decentralized systems: An overview," 3rd International Symposium on Autonomous Decentralized Systems (ISADS '97), Berlin, Germany, pp. 85-92, 1997.
- [352] Yoshiteru Ishida, Distributed and autonomous sensing based on the immune network *Artificial Life and Robotics: Computer Science and Engineering*, vol. 2, pp. 1-7, Mar, 1998.
- [353] Yoshiteru Ishida. Immunity-Based Systems A Design Perspective, Berlin, Germany: Springer-Verlag, 2004.
- [354] Yoshiteru Ishida and F. Mizessyn, "Learning algorithms on an immune network model: application to sensor diagnosis," *Proceedings of the International Joint Conference on Neural Networks (IJCNN'92)*, Beijing, China, pp. 33-38,

1992.

- [355] Yoshiteru Ishida and N. Adachi, "Active noise control by an immune algorithm: adaptation in immune system as an evolution," *Proceedings of IEEE International Conference on Evolutionary Computation (CEC'96)*, Nagoya, Japan, pp. 150-153, 1996.
- [356] Yuji Watanabe, Akio Ishiguro, and Yoshiki Uchikawa. Chapter 10: Decentralized Behavior Arbitration Mechanism for Autonomous Mobile Robot Using Immune Network. In: Artificial Immune Systems and Their Applications, ed. Dipankar Dasgupta. Germany: Springer-Verlag, 1999.pp. 187-209.
- [357] Zhou Ji and Dipankar Dasgupata, "Augmented negative selection algorithm with variable-coverage detectors," *Congress on Evolutionary Computation, CEC2004*, pp. 1081-1088, 2004.
- [358] Zhou Ji and Dipankar Dasgupta, "Real-valued negative selection algorithm with variable-sized detectors," *Genetic and Evolutionary Computation, GECCO-2004*, pp. 287-298, 2004.